The Effects of a Restorative Practice Program on Discipline Referrals, Discipline, Absences, and

Teacher Perspective in a Title I Middle School in East Tennessee

By

Nancy Harris Parker

A dissertation submitted is partial fulfillment of the requirements for the degree of

Doctor of Education

Education Department

School of Sciences and Education

Milligan College, Tennessee

2020

Doctoral Committee:

Major Professor

Dr. Mark Dula

Dr. Patrick Kariuki

Dr. Tausha Clay

@2020

Nancy H. Parker

All Rights Reserved

ABSTRACT

The purpose of this exploratory mixed-methods study was to investigate restorative practice initiatives in a Title I middle school in East Tennessee. Quantitative analysis was performed to look for significant differences between in-school suspensions, out-of-school suspension, alternative school, discipline referrals, and absenteeism before and after implementation of the program. Qualitatively, focus groups were conducted with teachers both before and after implementation of the program to gather perceptions of the program's effectiveness on student behavior and school culture as a whole. The quantitative data showed there was no significant difference in discipline referrals, out-of-school suspensions, absenteeism, and chronic absenteeism in fall 2018 and fall 2019. The number of in-school suspensions, combined suspensions (in-school and out-of-school), and alternative school referrals were significantly less in fall 2018 than fall 2019. The qualitative data found positive results from the focus group of educators with the implementation of this restorative practice program. The focus group reported students felt that trust and fairness played an important role towards their satisfaction with the restorative practice program. Also, the focus group reported the students developed good listening and communication skills and they felt a sense of empowerment to be a part of the initial restorative practice process. Based on the results of this study, further exploration of scheduling circle sessions, developing professional training for all teachers, and a longer period of time with the restorative practice program are needed to provide the data to see the success of this program.

Keywords: trust, fairness, communication, empowerment

Milligan College Dissertation Defense Approval Form

Candidate Name: Namey Parker
Date of Defense: $5/5/20$
Dissertation Title: The Effects of a Restorative Practice Program on
Discipline Refervals, Discipline, Absences, and Teacher Perspective
in a Title 1 Middle School in East Tennessee
Final Approval of Dissertation Defense
A signature below indicates committee members agree with the following.
 Agreement the dissertation meets with the committee's approval. Agreement the oral defense of the dissertation was successful.
Mark Dela 5/5/20
Dissertation Chair Signature Date
Patricle n. Karinkil 5/5/20
Dissertation Committee Member Signature Date
Dissertation Committee Member Signature 05/05/2020 Date
Dissertation Committee Member Signature Date
Patrick n. Karieken 5/5/20
Ed.D. Research Director Signature Date
Coula Hillan - figuration 05/05/2020
Area Chair of Education Signature Date

DEDICATION

This dissertation is dedicated to those most influential in my life. To my mother and father for instilling in me the inspiration to learn and for their understanding and loving hearts. To my daughter, who continues to fill my life with joy, love, laughter, and encouragement. They have inspired and helped me become the person I am today.

ACKNOWLEDGEMENTS

It is with sincere gratitude that I express my appreciation to the following individuals for their support, assistance, and encouragement in the writing of this dissertation. I praise God for giving me the strength, opportunity, and determination to pursue all my dreams. Dr. Dula, committee chair, thank you for your patience and support in the pursuit of my goal. Thank you for giving me the strength to persevere through this journey. Dr. Kariuki and Dr. Clay, committee members, thank you for your continued assistance and feedback during this project. Dr. Hilton-Prillhart, thank you for your insight and passion with this project. Thank you to all my professors at Milligan College who have inspired, supported, and mentored me throughout this process. I have been blessed to have great mentors who have taught me throughout this program. To the members of my cohort group, I would like to thank you for assisting me through this program and becoming lifelong friends. To my family, Mom and Dad, thank you for all your love, support, encouragement, and strength during this time. To my daughter, thank you for giving me joy in my life and being my beacon of hope. I would not have been able to pursue this dream without your encouragement.

TABLE OF CONTENTS

	Page
ABSTRACT	ii
DEDICATION	iv
ACKNOWLEDGEMENTS	V
Chapter 1: Introduction	1
Background Statement of the Problem Purpose of the Study Significance of Study Research Questions Definitions of Terms Delimitations Limitations Organization	3 5 7 8 11
Chapter 2: Review of Literature	13
Introduction Principles of Restorative Practices Progressive Discipline Approach Student Discipline and Restorative Practices Chronically Absenteeism and Restorative Practices Improving School Climate Through Restorative Practices Teacher Perceptions and Restorative Practice Positive Effects of Restorative Practices Summary	16 21 24 27 27 29
Chapter 3: Research Design and Methods	33
Introduction Rationale for Using Mixed-Methods Qualitative Component Quantitative Component Qualitative Research Questions Quantitative Research Questions Population	34 35 35 36 38
Sample	38

Instrument	38
Qualitative Instrument	
Quantitative Instrument	39
Data Collections and Procedures	
Data Analysis	41
Qualitative Strand	41
Quantitative Strand	43
Summary	44
Chapter 4: Data Analysis and Findings	45
Introduction	45
Demographic Data	
Qualitative Strand Findings	
Responses to Interview Question 1a	
Responses to Interview Question 1b	
Responses to Interview Question 2a	
Responses to Interview Question 2b	
Responses to Interview Question 3a	
Response to Interview Questions 3b	
Responses to Interview Question 4a	
Responses to Interview Question 5a	
Responses to Interview Question 5a	
Responses to Interview Question 6a	
Responses to Interview Question 6b	
Responses to Interview Question 7a	
Responses to Interview Question 7a	
Responses to Interview Question 8a	
Responses to Interview Question 8b	
Responses to Interview Question 9a	
Responses to Interview Question 9a Responses to Interview Question 9b	
Responses to Interview Question 10a	
Responses to Interview Question 10b	
Responses to Interview Question 11a	
Responses to Interview Question 11b	
Responses to Interview Question 12a	
Responses to Interview Question 12b	
Č	
Research Question 1 Findings	
· C	
Research Question 4 Findings	
Research Question 4 Findings	66
NESEAL O COESTION A PHICHIOS	On

Research Question 6 Findings67
Research Question 7 Findings
Summary69
Chapter 5: Summary of Findings
Purpose of Study70
Summary of Findings72
Quantitative Strand Findings72
Qualitative Strand Findings72
Discussion of Findings73
Quantitative Strand Findings73
Research Question 1
Research Question 2
Research Question 374
Research Question 4
Research Question 5
Research Question 6 &776
Qualitative Strand Findings
Research Question 8
Research Question 9
Research Question 10
Limitations of Study
Conclusions80
Recommendations
Recommendations for Practice
Recommendations for Future Research
REFERENCES87
APPENDICES98
APPREDIX A: Milligan IRB Approval99
APPENDIX B: Principal Permission to Conduct Study
APPENDIX C: Superintendent Permission to Conduct Study
APPENDIX D: Permission to Conduct Research
APPENDIX E: Focus Group Interview Questions (August)
APPENDIX F: Focus Group Interview Questions (December)

LIST OF TABLES

le Page	Table
. Teacher Responses to Interview Question 1a	1.
Teacher Responses to Interview Question 1b	2.
Teacher Responses to Interview Question 2a	3.
Teacher Responses to Interview Question 2b	4.
5. Teacher Responses to Interview Question 3a	5.
Teacher Responses to Interview Question 3b	6.
7. Teacher Responses to Interview Question 4a	7.
Teacher Responses to Interview Question 4b	8.
2. Teacher Responses to Interview Question 5a	9.
0. Teacher Responses to Interview Question 5b	10.
1. Teacher Responses to Interview Question 6a	11.
2. Teacher Responses to Interview Question 6b	12.
3. Teacher Responses to Interview Question 7a	13.
4. Teacher Responses to Interview Question 7b	14.
5. Teacher Responses to Interview Question 8a	15.
6. Teacher Responses to Interview Question 8b	16.
7. Teacher Responses to Interview Question 9a	17.
8. Teacher Responses to Interview Question 9b	18.
9. Teacher Responses to Interview Question 10a	19.
0. Teacher Responses to Interview Question 10b	20.
21. Teacher Responses to Interview Question 11a	21.

22.	Teacher Responses to Interview Question 11b	.60
23.	Teacher Responses to Interview Question 12a	.61
24.	Teacher Responses to Interview Question 12b	.61
25.	Paired Samples t-Test Comparing Fall 2018 and Fall 2019 Discipline Referrals	63
26.	Paired Samples t-Test Comparing Fall 2018 and Fall 2019 Out-of-School-Suspensions	64
27.	Paired Samples t-Test Comparing Fall 2018 and Fall 2019 In- School Suspensions	65
28.	Paired Samples t-Test Comparing Fall 2018 and Fall 2019 In-School & Out-of School	66
29.	Paired Samples t-Test Comparing Fall 2018 and Fall 2019 Alternative School	67
30.	Paired Samples t-Test Comparing Fall 2018 and Fall 2019 Attendance	68
31.	Paired Samples t-Test Comparing Fall 2018 and Fall 2019 Chronically Absent	.69

CHAPTER 1

Introduction

Background

Restorative practice is focused on the belief that those affected by harm can work together to repair the relationship and with a collaborative effort true accountability can occur. Wachtel (2016) states, "Restorative practice is a social science that studies how to build social capital and achieve social discipline through participatory learning and decision-making." Restorative justices are defined as a subset of restorative practices that consist of formal or informal responses to crime and other misconduct after it occurs. The definition of restorative practices will include the use of informal and formal processes that precede the offense, and proactively build relationships and a sense of community to prevent conflict and wrongdoing (Wachtel, 2016). The concepts of reparation and restoration are integral to understanding restorative justice. Much of the diversity in practice and definition is due to the fact that restorative justice theory has developed alongside restorative justice practice (Daly & Proietti-Scifani, 2011).

According to Howard (2009), the use of restorative practice in schools is a fresh idea for education. The application of the longstanding doctrine of restorative fairness has been embraced by police forces in North American, England, and Australia. The police have applied restorative impartiality to a wide range of situations from immediate responses to complex situations, regarding the harm of individuals with very serious consequences. In recent years, restorative practice has not been restricted to the justice system entirely, but has been used in the most multifaceted political and community state of affairs. Restorative justice was the root for

the peacemaking practice that ended the civil war in Papua New Guinea and provided another example of restorative practice in the mid-1990's following the official ceasefire in Northern Ireland where there was growing interest in developing a non-violent, restorative community as an alternative to the ongoing use of retribution and other acts of vengeance. Howard (2009) stated despite the intricacy and continual nature of the differences these large-scale attempts to affect restorative practice had a major impact.

From 1974 to the mid-2000's, suspension and expulsion rates doubled in the United States (National Center for Education Statistics, 2009). A study out of Texas found that across three cohort groups of seventh graders, more than 900,000 students over the course of six years, nearly six in ten public school students were suspended or expelled at least once between their 7th and 12th grade school years (Fabelo et al., 2011). A move away from zero tolerance policies is currently decreasing the number of suspensions nationwide (U.S. Department of Education, 2016). "In 2008, the American Psychological Association 's Zero Tolerance Task Force found schools with higher suspension and expulsion rates had lower ratings for overall school climate." This research indicated that suspensions and expulsions did not reduce future student misconduct. Restorative practices are one possible alternative to zero tolerance policies.

According to Skiba, et al. (2014), restorative practices promote a constructive and orderly school environment. Students and all members of the school population can learn and practice self-discipline, compassion, and accountability for their actions in all situations. A restorative practice program incorporates restorative circles that bring students involved in the disciplinary actions together in deferential space to promote conversation, provide accountability and resolve all conflicts between the community stakeholders. Restorative practices can be applied in a variety of ways so schools, families, and populations from lower income

communities can provide a means to their educational system that can be inexpensive to implement but have great significance in their school system (Skiba et. al., 2014).

In the last few years, there has been a significant interest in the application of restorative practice philosophy within the educational culture as a whole. According to Zehr (2002), restorative practices are based on the concept that when disagreement occurs and one or both participants are harmed, the damage needs to be addressed. Restorative practices advocate ownership of the behavior and involves discussions where accountability for the problems are established through a consensual agreement. Restorative practice relies on the basic belief that people are connected through a network of relationships and when harm occurs between people the community is also harmed. Educational institutions across the country are encouraged to implement restorative practice programs as an alternative to suspensions and other disciplinary actions to replace punitive techniques to discipline. Restorative practice programs can significantly improve the school environment and support the social and emotional abilities of students and teachers. Many studies have provided an emergent awareness to interventions, such as restorative practice programs, that help students recognize the effects of their actions. As a result of restorative practice programs, studies have reported lower suspension rates, improved school climate, and improved student attendance.

Statement of the Problem

In light of research, punitive discipline methods were far from effective in helping students reach their goals. Punitive discipline resulted in a poor school climate, attendance and almost doubled the number of suspensions in the 1970s (Boucaud, 2019). Restorative practice will identify the difference in discipline, attendance, and teacher perspectives pertaining to the restorative practice approach in a Title I middle school in East Tennessee before and after

implementation. Skiba (2014) suggested "Exclusionary policies have a negative impact on school climate and students' engagement with school." Recent studies indicate discipline practices, such as suspension, might be hindering student success. One study found suspension to be the top predictor of students dropping out of school (Flannery, 2015). Another report estimated graduation rates for suspended students were 68 percent, compared with 80 percent for non-suspended students (Rumberger and Losen, 2016). Often these suspensions involve non-violent offenses (Skiba et al., 2014). Violations include tardiness, absenteeism, and disrespectfulness to instructors (Gonzalez, 2012). A study by Gonzalez (2012), suggests restorative practices can both prevent harm through building relationships and stakeholder involvement to repair damaged relationships.

In many schools, statistics show punitive strategies like suspension and expulsion are not working. In one recent study, schools with high suspension rates were found to be are less safe and less equitable (Skiba et al., 2014). Zero- tolerance policies may seem like the answer to disciplinary actions; however, in the short term, these policies emphasize punishment and its consequences instead of examining the root cause of student behavior. Additionally, these policies do little to repair the damage to relationships and fail to prevent recurrence. Restorative practices in a school setting are generally seen as a non-punitive technique to handling conflict (Fronius et al., 2016).

Purpose of the Study

The purpose of this exploratory mixed-methods study was to examine the effects of the implantation of a restorative practice program on discipline, discipline referrals, absenteeism, and teacher perspectives. The school in this study was a low-income middle school in East Tennessee and was considered a Title I school. The U. S. Department of Education's purpose of

Title I funding "is to ensure that all children have a fair, equal, and significant opportunity to obtain a high-quality education and reach, at minimum, proficiency on challenging state academic achievement standards and state academic assessments" (U. S. Department of Education, 2013). This school received free lunch for all students at the beginning of the 2019-2020 school year. Teacher perceptions, absenteeism, and disciplinary measures both before and after the implementation of a restorative practice program will identify any positive and negative changes.

As part of the mixed-methods study used for this research, focus group interviews with teachers and quantitative methods were used to gain information on ISS, OSS, alternative school, discipline referrals, absenteeism, and teacher perspectives. Two interviews were conducted with a focus group of teachers regarding the perceived needs for the program prior to the study and again after the program was initiated. Data was obtained through PowerSchool regarding ISS, OSS, alternative school, and absenteeism. Discipline referrals were used to generate teacher discipline reports. After the study is complete, findings will be shared with the teachers and administrators that were involved in the study.

Significance of Study

This study examines a restorative practice program before and after implementation in a low-performing Title I middle school in East Tennessee. Restorative practices show promise with decreasing discipline referrals, suspensions, and expulsions. According to Gregory et al. (2016), over the course of a school year, greater use of restorative practices was associated with lower teacher referrals for misconduct and student behavioral issues. Another report conducted two years after launching restorative practices in an Oakland, California middle school found a 74 percent drop in discipline referrals and 77 percent drop in referrals for violent behavior

(Summer, Silverman, and Frampton, 2010).

Some studies have reported links between implementing restorative practices and improved overall school climate to include up to three years after implementation (Jain et al., 2014). In Oakland, California, two thirds of school staff reported in a survey that they acknowledged the restorative practice program as having improved the social-emotional growth of students (Jain et al., 2014). Some studies have reported increased openness and connectedness between students and teachers and greater respect for students after implementing restorative practices (Gregory et al., 2016). Another study indicated that students had enhanced problem-solving skills after exposure to restorative practices, indicating they could better manage conflict with each other and with school staff (McMorris et al., 2013).

Other related studies have looked at the connection between restorative practices and student attendance, which have found positive correlations. One study found chronic absenteeism in schools implementing restorative practices decreased by 24 percent as opposed to schools who did not implement the program where absenteeism actually increased by 52 percent (Jain et al., 2014).

This study will provide findings of restorative practice initiatives introduced in a Title 1 middle school in East Tennessee. This study also seeks to identify techniques to guide schools in their implementation of restorative practices as well as to provide feedback to educators regarding teacher's perspectives on restorative practices. Additionally, the study will use t-tests before and after to examine discipline referral patterns and attendance following implantation of a restorative practice program.

Research Questions

Several questions were formulated in order to examine the implementation of restorative practices in a Title 1 middle school in northeastern Tennessee. The questions examine teacher perceptions regarding restorative practices and the relationship between implementation of a restorative practice program and disciplinary actions, attendance, and school climate. The research questions for this study are as follows:

RQ1: Is there a significant difference in the number of student discipline referrals during fall 2018 semester versus fall 2019 semester, following the institution of restorative discipline practices?

RQ2: Is there a significant difference in the number of out-of-school suspensions during fall 2018 semester versus the fall 2019 semester, following the institution of restorative discipline practices?

RQ3: Is there a significant difference in the number of in-school suspensions during fall 2018 versus fall 2019 semester, following the institution of restorative discipline practice?

RQ4: Is there a significant difference in the number of students suspended (both in-school and out-of-school) during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

RQ5: Is there a significant difference in the number of students placed in an alternative school during fall 2018 semester versus fall 2019 semester, following the institution of restorative discipline practices?

RQ6: Is there a significant difference in the number of absences during fall 2018 semester versus fall 2019 semester, following the institution of restorative discipline practices?

RQ7: Is there a significant difference in the number of students who are considered "chronically absent" during fall 2018 semester versus fall 2019 semester, following the institution of restorative discipline practices?

RQ8: What are the perceptions of effectiveness, fairness, and training/satisfaction of teachers after training (but before institution) in restorative discipline practices?

RQ9: What are the perceptions of effectiveness, fairness, and training/satisfaction of teachers after a semester of instituting restorative discipline practices?

RQ10: Have teachers' perceptions of effectiveness, fairness, and satisfaction changed regarding restorative discipline practices in this time frame?

Definition of Terms

Restorative Justice: Understanding the definition of restorative justice is crucial for discernment of this study. Braithwaite (1999) defines restorative practices as those that promote healing rather than hurting, community involvement, respectful conversation, forgiveness, and making reparations. Hopkins' (2003) defines restorative practices as practices focused on managing behavior and shifting away from punitive measures. Fronius et al. (2016) suggest that restorative justice practices are broadly described as non-punitive approaches to handling conflict. Restorative justice is defined by its specific emphasis on respect and relationships. By upholding members' dignity and addressing their human needs, the process encourages and gives power to partakers to discuss their encounters openly and honestly and to collaboratively settle on reparation.

Suspension: Suspension is a compulsory leave handed over to a student as a type of punishment that can last anywhere from one day to a few weeks, during which the student is not allowed to attend regular classes or participate in school activities (Skiba, 2006). In schools in the United States, there are two types of suspension; In-School (ISS) and Out-of-School (OSS). In-school suspension means the student comes to school during the regular school day but reports to a designated room or area for the entire school day (Skiba, 2006). Out-of-school means that the student is barred from entering the school or school grounds (Skiba, 2006). Alternative school means the students are placed in an alternate classroom setting at a designated location for a given length of time based on the discipline policy of the school system. Students who do not adhere to the suspension could be arrested for trespassing and it could result in an extension of their prior suspension. Students are not allowed to attend any after-school activities while they are suspended from school (Skiba, 2006).

Discipline Referrals: A discipline referral is a way for a teacher to refer a student to an administrator to be disciplined. It's usually a physical form that a teacher will fill out and turn into the office with information about the occurrence and what steps the teacher has taken to correct the incident. An administrator will be given the referral and decide how to officially discipline the student. A referral means the misconduct is either too severe for the teacher to properly discipline the student in the classroom or the teacher has tried to discipline the student on his or her own without success. More serious problems, such as drugs and violent behavior, should always mean a discipline referral and instant removal from the classroom. Others within the school environment may write discipline referrals such as coaches, counselors, bus drivers and other educational personnel if they observe a behavior that needs to be reported.

Chronically Absent: Chronically absent is defined as a child enrolled in a school under the authority of a local board of education whose total number of absences, at any time during a school year, is equal to or greater than 10 percent of the total numbers of days a student is enrolled during the school year.

In Tennessee, the law measures each school district by what is called the Chronically Out-of-School Indicator. This indicator will be based on what research calls chronic absenteeism, which is defined as a student missing 10 percent or more of the days the student is enrolled-for any reason, including excused absences and out-of-school suspensions (Tennessee Department of Education, 2019).

Absence: Absence means an excused absence, unexcused absence, or disciplinary absence, defined by the State Board of Education. In Tennessee the policy states absent is when a student is marked not present at school, for either unexcused or excused reasons. According to the state attendance policy a student who is absent for half of a school day should be marked absent. Excused absence is a student who is absent due to excused reasons which usually requires documentation as defined by the individual school district. An unexcused absence is a student who is absent due to unexcused reasons and each school district will define the policy stating what constitutes an unexcused absence.

<u>Teacher Perception:</u> Teacher perception refers to the teacher's own distinctive experience in the restorative justice program with regards to the effectiveness, fairness, satisfaction after training (but before institution) in restorative discipline practices and after a semester of instituting restorative discipline practices within the classroom. Teacher perceptions will be measured qualitatively through the focus group interviews.

Delimitations

The restorative practice program was put in place during the fall of 2019 and all seventh and eighth grade students' data were applied in this study. Data used in this research was discipline referrals, ISS, OSS, and alternative school information and absentee reports. This study was conducted in a single middle school in upper east Tennessee.

Limitations

To ensure reliability, PowerSchool information and data regarding ISS, OSS, and alternative school data along with absentee reports must be accurate and up to date. Student behavior and absenteeism information could be determined on demographic and the student's socio-economic environment. The teachers involved in the focus group, answered interview questions pertaining to their perspectives on the implementation of the restorative practice program. All information was based on their personal responses to the perceived comfort with restorative practice programs. Potentially, the focus group's perspectives will vary due to the academic and social differences of their classroom students.

Study Organization

This dissertation is structured into five chapters. Chapter 1 introduces restorative practices involving ownership of behavior and discussions with participants where accountability for the problems is established through a consensual agreement. A Title I middle school began the restorative practice program to improve discipline, attendance, and school climate. The statement of the problem, the purpose of the study, the significance of the study, definition of terms, delimitations, and limitations of the study are made available.

Chapter 2 provides a review of literature involving restorative practice programs and cites several case studies with regard to zero tolerance, chronically absenteeism, student discipline, and progressive discipline approaches. Information is provided on improving school climate with the use of restorative practice programs. Finally, Chapter 2 provides information on the positive effects of a restorative justice program within other similar school case studies.

Chapter 3 delineates the methodology for this study. The research plan and justification are addressed in this chapter. The data compilation procedures and methodical techniques for both the qualitative and quantitative data research component is presented. Chapter 4 presents the results of this study. Chapter 5 provides a dialogue of the outcomes and implications for prospective study.

CHAPTER 2

Literature Review

Introduction

Restorative practices in schools started with the International Institute for Restorative Practices (HRP) in the 1990s. HRP began developing SaferSanerSchools to intervene in American education and society as a whole (Chmelynski, 2005). Wachtel (2016) states there is a remarkable change in student behavior due to the loss of attachment to parental and community involvement in today's culture. Schools have generally become larger, more impersonal, and educators feel less connected with the families of the children they teach (Chmelynksi, 2005). Restorative practices involve changing relationships by engaging people to participate in activities together rather than participate in activities by themselves or for themselves. Research suggests a foundation of restorative practice may replace of out-of-school suspensions to change behaviors, lower suspensions, and increase attendance.

Traditional measures involving suspension are largely ineffective according to research (American Psychological Association, 2008). School districts are looking for alternative ways to create an environment of safety, accountability, and fairness. Many school districts are encouraging schools to introduce restorative practices in place of disciplinary punishment.

Restorative practice programs vary in design but most bring together those who were harmed and those engaged in the harm with adults representing the school and community. For example, McCold (2003) states restorative circles and restorative conferences allow victims, offenders and other stakeholders to come together to explore how everyone was affected by the offense and how to repair the harm and meet their needs. For educators, circles provide opportunities for students to share their feelings, build relationships, solve problems, and to play an active role in

addressing disciplinary action (Riestenberg, 2002).

Schools are affected by student misbehavior. Research shows that many of these depraved behaviors from students are an attempt to fulfill a genuine physical or psychosocial need or to attain an adult's attention (Haiman, 1998). According to Solter (1998), many incidents of misbehavior are signs of children or youth trying to meet their own needs or to get the attention of an adult. "Obvious needs are for food, clothing, safety, shelter, and love. Additional needs are for physical closeness, individual attention, an intellectually stimulating environment, meaningful activities, and respect." Research indicates that because children often don't know how to ask for help or have not developed positive strategies to fulfill their needs, they can turn to negative behaviors (Solter, 1998).

Historically, inappropriate behavior has primarily been addressed through punitive disciplinary approaches. The majority of educational districts have a zero-tolerance policy for unacceptable school behavior. To ensure a safe and secure learning environment, many district policies state any student who engages in drugs, weapons, aggravated assault on a staff member or other employees of the school on school property or at a school activity, and any transmissions of a threat by electronic devices will result as a zero-tolerance behavior. All discipline for a zero-tolerance offense as described by the districts usually results in expulsion for a period of not less than (1) one calendar year. The Director of Schools for each district is given the authority to modify these expulsion requirements on a case-by-case basis as part of the appeal procedure set out in each prescribed Student Rights and Responsibility Policy for the school district. Research has shown disciplinary practices such as suspensions, expulsions, and zero-tolerance policies are damaging and do not make schools safer (Haft, 2000). Such disciplinary measures weaken school relationships and hinder students' reentering into the

school environment (Bazemore, 1999). Diminishing school climate using these practices of zero-tolerance has caused some to say it is contributing to what some call a school to prison pipeline in the United States. The school to prison pipeline is a process where students are pushed out of schools into prison. This practice of criminalizing youth is carried out by disciplinary policies within schools that put students in contact with law enforcement (Cole, 2019). "Once they are in contact with law enforcement for disciplinary reasons, many are pushed out of the educational environment and into the juvenile and criminal justice system," (Cole, 2019, p.1).

A goal of every school should be to understand how restorative practices can change the students' behavior towards building positive relationships between themselves and their teachers, thus improving the school climate. One major factor to discuss is the difference between discipline and punishment. Arnold (2014) states school discipline relates to the actions taken by a teacher or school association towards a student (or group of students) when the student's behavior interrupts the current educational activity or breaks a rule created by the teacher or the school system. It describes the term "discipline" as pertaining to the punishment that is the result of breaking the rules while the goal of discipline is to set restrictions limiting certain behaviors or positions that are seen as harmful or against school policies and educational standards. Today many school systems are shifting the focus from punishment to alternative approaches such as restorative practices to offset high dropout rates and inequitable punishment upon minority students (McCluskey et al., 2008). Restorative practice is a model that has been used with some success within juvenile justice and education programs. The model looks for a way to repair harm to others by recognizing the impact on the victim, community, and offender. It provides an avenue of accepting accountability for the misconduct and repairing the relationship with the victim. In schools, restorative practices might include teachers and students devising classroom

expectations together or setting up community building in the classroom (Davis, 2015). Zero tolerance policies have not always worked well for students across the cultural and social community (Davis, 2015). As an alternative to the policies of corporal punishment, suspension, and expulsion, restorative justice programs were launched to give students a voice in their discipline, as well as an opportunity to make a constructive contribution to their community (Davis, 2015). Each student who has misbehaved has the opportunity not only to give their side of the story but also to have a voice in determining their consequences. Consequences focus on opportunities for restoration (Dalporto, 2013). The focal point of restorative justice is relationship building within the community over the individual student and their delinquency, creating awareness that everyone has a part in the community, and it is everyone's responsibility to support the values of the community (Dalporto, 2013). In turn, a restorative justice program not only increases an understanding of the community's values but also provides a relationship to be developed between the community and the school.

Principles of Restorative Practices

In recent years, restorative justice approaches have been making a strong presence in schools across Canada and the United States as zero-tolerance policies concede to a more non-adversarial approach to discipline and school climate matters. Restorative practices have been described as "a whole-school ethos or culture comprising principles and practices to support peacemaking and solve conflict through healing damaged relationships and making amends where harm has occurred while preserving the dignity of everyone involved" (Meyer & Evans, 2012). Restorative practices are positive as they aid in constructive relationships and school climate. Meyer and Evans (2012) state several principles of restorative practice should include the following: Interpersonal relationships which are healthy interpersonal relationships

developed within the school community with students, teachers, parents, personal dignity which preserves the idea that every person belongs and is respected, cared for and has the right to equitable treatment, mutual respect and understanding where a respectful conversation about what happened is established and a value is given for each person's experience, restorative conferencing where a pledge is made to resolve a situation using conflict resolution and healing of interpersonal relationships, and restitution where an agreement is established and efforts made to mend the harm, resolve conflict and reinstate interpersonal relationships.

Under the definition of restorative practice, there is no single standard of practice because each school has its group of people, history, strengths, and areas of improvement that must be reflected in their approach. While there will be differences in the types of restorative practices implemented, there is one common factor among all of them which is the opportunity for the victim and the transgressor to discuss how they are involved, to share their experiences honestly and safely, to take responsibility and work in partnership to determine restorative consequences to repair the harm (Drewery, 2004).

Restorative practices range from informal to formal activities, which involve stakeholders (Wachtel, 2016). These stakeholders will include students, educators, families, and local communities (Restorative Practices Working Group, 2014). A list of activities should include the following: (1) Community conferencing, which involves students and educators, affected by the misbehavior to participate in the conflict decision and prevention process. (2) Providing a community service program that requires the student to complete meaningful and relevant service to repair damage to the school community. (3) Peer juries which consist of trained student jurors to discuss with the student the reason for their misbehavior and to determine who was affected and provide restitution. (4) Circle processes are common techniques used to foster

relationships and a sense of school and community pride to respond to bad conduct. This technique of restorative practice develops interpersonal and communication skills by inviting students and educators to take a turn speaking (i.e. about what occurred, how they feel about it, how it affected them and the school along with how to make amends) and listening in a controlled environment. (5) Preventative and post-resolution programs are tools that provide students with opportunities to learn problem-solving and instructional skills. (6) Peer mediation where a student leadership genre is used to train student leaders to support their fellow students in resolving conflicts. Affective statements and questions are informal techniques used to make possible communication of feelings and expressions on the impact of behavior for students who misbehave. Social-emotional learning allows students to learn skills that enhance their awareness and management of emotions, to develop empathy and to build positive relationships (Restorative Practices Working Group, 2014; Wachtel, 2016).

Restorative practices exemplify an attitude that, when infused into the foundation of the school community, can be engaged to extend and sustain a safe and caring school climate.

According to the International Institute for Restorative Practices, restorative practices "reduce crime, violence and bullying, improve human behavior, strengthen human behavior, strengthen civil society, provide effective leadership, restore relationships and repair harm" (Wachtel, 2016). Additional research from the International Institute for Restorative Practices found fewer disciplinary referrals, reduced rates of suspensions and expulsions, less instructional time lost due to behavior management, improved educator morale and retention, and improved academic outcomes and a decline in disproportional rates of disciplinary referrals of minority students (Porter, 2007). Restorative practices are also associated with improvements in attendance, academic achievement and graduation rates, and a reduced number of classroom disruptions thus

improving the entire school climate (McMorris et al., 2013).

Additionally, restorative practices may serve as an alternative to other disciplinary procedures. Zero-tolerance policies consist of viewpoints related to punishment originally intended for drug offenses in the 1980s (Gonzalez, 2012). The war on drugs went from the streets to the schools, where both severe offenses (such as bringing a weapon to school) and minor offenses (such as cussing a teacher) were not tolerated on school property (Gonzalez, 2012). In the 1990s, these policies were put into practice in school districts to target and prevent violent behavior (Winslade & Williams, 2012). The U.S. and Canada passed legislation for school systems to establish operation policies that were strong on crime (Evans & Vaandering, 2016). The U.S. passed the Gun-Free School Act in 1994, and Canada adopted the Safe Schools Act in 2000. These guidelines have shaped how school environments react to infractions within their community by emulating the conventional criminal justice approach (Gonzalez, 2012).

Gonzalez (2012) states that several research studies suggest punitive discipline policies in school both exclude students from educational opportunities and fail to make schools safer.

Also, students have a higher chance of becoming part of the juvenile justice system when zero-tolerance policies disconnect them from their school and community.

Progressive Discipline Approach

A study in Ontario, Canada was enacted in 2008 when the Ontario Education

Commission regarding behavior, discipline, and safety introduced a bill. The proposed

legislation called for a more progression disciplined approach such as prevention and early

intervention on student misbehavior. The law also provided a program solution for successful

reentry of suspended and expelled students. Progressive discipline approaches rely on a variety

of disciplinary interventions so students can learn from their mistakes and improve their

behavior. The "framework of progressive discipline shifts the focus from one that is punitive to a focus that is supportive and corrective in nature" (Ontario Ministry of Education, 2006). Progressive tactics of discipline include meetings with students and their parents, taking away privileges, peer arbitration, and restorative practices. The shifting away from zero-tolerance towards progressive discipline measures in education came from the criminal justice system. Restorative practices have their roots in the cultural customs of people around the world who seek to "live and learn in a good way with each other" (Durham District School Board Safe Schools, 2012) and value "harmony and respect" (Lewington, 2016). Although this practice has been around for years, its modern-day approach comes from restorative justice or other alternative methods to discipline in the criminal justice system. Correctional institutions in Canada have defined restorative justice as a "non-adversarial, non-retributive approach to justice that emphasizes healing in victims, meaningful accountability of those responsible for the harm and the involvement of citizens in creating healthier, safer communities. "Restorative justice works to repair the damage and promote healing and growth" (Pakan, 2007). The Canadian institutions suggest that traditional methods of punishment do not emphasize repairing harm and restoring respect while developing relationships with the victim, as the restorative justice approach does.

Ontario's move toward making schools safer involved the entire school community to center on collaboratively promoting constructive student behavior through a policy of regulations, structuring healthy and courteous relationships throughout the whole school community, averting inappropriate behavior through prevention programs, and concentrating on inappropriate behavior through progressive discipline methods such as restorative justice (Pakan, 2007).

Student Discipline and Restorative Practices

Many school districts are looking for a different solution to the traditional punitive model of school discipline. The districts want a distinctive solution that yields better results in terms of resolving conflicts and preventing students from repeating their offenses. Restorative justice programs have been well researched and are considered applied methods that help educators to rethink the disciplined approach (Catapano, 2019).

In your typical school model of discipline, behavioral infractions earn a harsher penalty for each progressive incident of the behavior. The biggest problem with this type of model is that a wide range of behaviors receive the same punishment. Catapano (2019) sites an example involving a student who receives multiple tardies receiving detention in the same way as a student who swears at a teacher. "The Civil Rights Project, a division of UCLA research, reported that 3.3 million K-12 students are suspended each year, and 102,000 are expelled." If a different approach to discipline could be considered these extreme disciplinary measures could be avoided. Many schools maintain a thin array of disciplinary options that fall short of the full range of components considering the student's actions. Catapano stated in 2007, the "National Center for Education Statistics reported that nearly one in every four students nationwide had been suspended at least once."

Restorative school discipline should not be considered an add-on program for behavior management or another tool for educators to use to deal with student behavior. Restorative justice programs and school discipline should represent a school culture that permeates all parts of the school's organizational atmosphere, the associations within the school, as well as interactions between the school and community. Ultimately, restorative practice programs recognize schools are educational institutions, and strategies and procedures should be for the

students and their school community. School districts should bear in mind responses to student behavior should be consistent with their educational goals of supporting teaching and learning and not punishment, retribution, and exclusion (McCluskey et al., 2008).

The school should be the vehicle where every student's behavior should be consistent with the educational goals of the school community and are dealt with through compassionate educational involvement.

Restorative school discipline could be the basis to educate not just in one subject area only but beyond to include citizenship and become a contributing member of the student's community. Restorative justice discipline programs could provide all members of the school community with the skills for constructive social interactions, associations, and decisions that support educational learning.

One recent study provided by the Chicago Public Schools reported that due to changes in school discipline policies, training in classroom management, and the use of restorative practices on school discipline issues with teachers and students resulted in, according to one teacher, "lawlessness" (Perez, 2015). Several schools report that they did not have a behavioral specialist or a designated school staff member to intervene with students, staff, and parents on discipline issues nor the resources to train staff members on discipline practices that address the students' needs. District training on restorative practices was not provided to all schools and the restorative practice program was only used on a weekly basis. The study suggested the new discipline policies enacted by the district placed more substantial limits on the use of suspensions and sought to keep students in the school environment.

Knoff (2015) advises that restorative practice programs are beneficial but "only when they are matched to the students who will most benefit from those programs." He suggests

schools should focus on programs like restorative practice, which reinforces the skills students need to cope with regarding behavioral issues. He felt a restorative practice program did provide intervention strategies to help the most challenging students tackle their unacceptable behaviors (Knoff, 2015).

Classroom management continues to be an issue for every educator, administrator, and parent. Educators should focus on providing a safe and supportive classroom that allows students an opportunity to learn and grow. School districts have several frameworks and guidelines that are used for stopping and preventing student behavioral problems. Suspension is one of the options used in disciplinary measures. School districts have both in-school suspension (ISS) and out of school suspension (OSS). These suspensions are administered because a student is severely disrupting the learning atmosphere. From a school personnel perspective, the only recourse is to remove the offending student so threats to the safety of students and school personnel are limited. Research and surveys have suggested that the most cited discipline problems have little to do with violence but instead focus on insubordination and defiance (Skiba, Peterson, & Williams, 1997). A study of a large school district in Florida also found insubordination and other non-violent offenses compromised the majority of OSS (Mendez et al., 2002). Skiba, Peterson, and Williams examined the disciplinary referrals and punishment of two middle schools in the Florida district and found that the majority of offenses led to non-violent referrals. In-school suspension became a subsidiary of the OSS plan because many educators were frustrated with the discipline design and felt that ISS was a more rehabilitative depiction of discipline. ISS keeps students in the classroom atmosphere, and school officials can punish inappropriate behavior more constructively. There are often limited case models of school districts that use ISS so effectively that if dramatically changes the disciplined environment and

suspension rate in their schools (Haley & Tomczyk 2000). Turpin & Hardin (1997) felt ISS programs were a limited success and did not alone validate the effectiveness of this method for school discipline.

Chronic Absenteeism and Restorative Practices

The behavior of absenteeism is described as a period when a student is not attending school, which in turn often leads to the loss of academic success (Strickland, 1998). Although not all students who are chronically absent are considered truant, the non-attendance of school can be detrimental to students in many other ways (Strickland, 1998). For example, perfect attendance does not determine grade point average, but lack of attendance does have a relationship to grade point average (Strickland, 1998). Attendance is a problem in the United States and students are missing more school than they attend (NCTQ, 2014). If a student is chronically absent or truant, suspending them from school only continues the cycle of behavioral issues and mitigates their ability to come to school to gain the academics required for graduation. Research indicates students missing more than 10% of school during a school year are more likely to drop out (Balfanz & Byrnes, 2012).

Improving School Climate Through Restorative Practices

A positive school climate means everyone who includes students, parents, staff and community members feel they are in safe hands and appreciated. Everyone has a role to play in promoting healthy relationships and a school climate that encourages appropriate student behavior.

One study at Cole Middle School in Oakland, California, began a program of restorative practices in 2005-06 school years. Significant benefits in terms of quantifiable data were noticeable between 2006-07 and 2007-08 school years when the suspension rate at the school

dropped. The school had been suspending almost one-third of the student population (30.3%). After the implementation of the restorative practice program, student suspension rates dropped to 10.3% (Jain, et al., 2014). Suspensions declined by 87%, and expulsions fell to zero during the implementation of who school restorative practice (Sumner et al., 2010). The staff reported that restorative practice programs improved the overall school climate during the first year of implementation (Jain, et al., 2014). The students shared the use of restorative justice circles enhanced their ability to understand their peers better, to manage their emotions, to develop greater empathy, and to resolve conflict with parents, improve the home environment, and maintain positive relationships with peers (Jain, et al., 2014).

Another study conducted in public schools by Port, et al. (2012) found there was significant progress made with student behavioral issues and a positive school climate atmosphere was present as a result of the reparative dialogue established through restorative practice programs. Grossi and Santos (2012) also found students demonstrated increased respect for other students while showing more promising approaches to conflict. Additionally, restorative practices played a primary role in helping to resolve disputes in the school, which provided a better overall school climate.

Researchers found positive results for the school climate of Midway High School in New Zealand through the use and practice of restorative programs. Kaveney and Drewery (2011) reported teachers using restorative practice programs had better relationships with their students. An improvement in student awareness was found, and teachers reported restorative practices had positive effects on teacher and student relationships. The school climate improved with students being more caring, respectful, and considerate toward the staff and fellow students.

McCluskey (2008) reported positive results from a two-year pilot program, which

investigated the effects of restorative practices in eighteen schools in Scotland. Findings in the research showed improvement in school culture and a decrease in discipline referrals. In Adelaide Australia at Lonsdale Primary School, administrators implemented a restorative justice program to offset increasing aggression within the school environment. Lane (2005) reported that not only an improvement in school climate was apparent, but also parents were more supportive of restorative based programs versus other disciplinary techniques.

Restorative practices were introduced at Palisades Middle School in Pennsylvania during the fall of 2000. According to Mirsky (2003), the school climate was ill-mannered and impertinent and disagreements were common. The school principal reported he was suspending around 200 students during a school year for everything from disrespect to skipping classes. The entire staff was trained in restorative practices. The teachers used restorative practices in the classrooms in the form of circles, where students and faculty share information and problems. Students wrote in their journals about what happened in a discipline situation and suggested how to resolve the situation. The school saw a statistical decrease in the number of problems that occurred each day and data indicated a substantial drop from the 2000 school year to the 2001 school year in discipline problems and incidents of fighting. Teachers felt restorative practices had a positive effect on academic performance while providing an improvement in the school climate (Mirsky, 2003).

Springfield Township in Philadelphia received many urban transfer students who moved from place to place regularly. The administration felt their biggest challenge was integrating students from different systems into the culture of the school (Mirsky, 2003). The staff was trained in restorative practices and used check-in and checkout circles with both their classroom management and academic issues to help create the culture deemed appropriate for the school

environment. The administration provided a list of questions for students to think about during their in-school suspension assignment. These questions asked the student to think about: "What happened? Who do you think has been affected by your actions? What can you do to repair the harm?" The administration then discusses the questions with the student and brings in the teacher involved to talk about the situation and mend any disputes. The administration considers restorative practices to be a part of the building culture and data shows that since restorative practices were introduced a decrease in discipline problems has occurred. The improvement in the school culture and climate have changed the school's core values with restorative practices (Mirsky, 2003).

Teacher Perceptions and Restorative Practice

Studies suggest punitive measures like those currently used in our justice system do not reduce recidivism or decrease violence in youth (Choi, Green, & Gilbert, 2011). Schools have shown they are not prepared to deal with the increasing amounts of violence, bullying, and other serious problems affecting our communities (Morrison, 2006). Behavioral issues and other aggressive acts are significant concerns for educators since student success depends on how well students interact with their peers and adapt to the school environment (Alvis, 2015). Ineffective disciplinary methods in schools have helped to perpetuate aggression acts resulting in the rising rates of suspensions and detentions of students (Morrison, 2006). School experiences help students gain the emotional skills needed to engage and interact positively with their peers (Goleman, 1998). For this reason, schools need to examine alternative means to discipline that will help prevent hostile behaviors and help students build better relationships with their teachers and the school environment.

In recent years, restorative justice supporters have begun promoting restorative practices

in schools. Restorative justice approaches allow students to understand the effect of their negative behaviors on those that have been harmed and help them to restore relationships while making amends (Zehr, 2002). In schools using restorative practice programs, therapeutic interventions are used instead of more punitive methods to behavior management. Restorative practices work from a whole-school, strength-based model that allows for meaningful and supported opportunities for students to take responsibility and be accountable for their actions (Zaslow, 2009). Students and teachers work together by discussing inappropriate student behaviors and collaboratively reaching an agreement that meets the needs of everyone involved (Varnham, 2005). Restorative activities include peer mediation, restorative conversations, classroom circles, thinking plans, small and large group restorative meetings, and formal restorative conferences (Macready, 2009).

Teachers use restorative practices at the elementary level to help instill values, skills and ethical principles for working with others (Morrison & Vaandering, 2012). At the secondary level, educators address specific behaviors that disrupt the harmony of the school and classroom environment through problem-solving circles, conferences, and peer mediation. Morrison and Vaandering (2012) found that restorative programs increase positive emotions such as empathy and while diminishing negative emotions such as anger, humiliation, and fear. Hargreaves (1997) proposed that emotions, feelings, relationships, and human interactions all influence learning. Teachers found that students who participate in restorative circles form a different kind of relationship with each other that is based on inclusiveness, empathy, equality, connectedness, and respect (Boyes-Watson, 2005).

A study was conducted in a Title 1 public suburban elementary school in northeastern New York with over 1000 students in Pre-K to 6th grade. The study found that teachers felt the

major strengths were connections made by the staff between all stakeholders, had a set of standard procedures allowing stakeholders to know what to do, expect, and made a great impact on school climate.

Positive Effects of Restorative Practices

Many school districts, along with their stakeholders have become increasingly concerned about suspensions especially about suspending younger children from elementary and middle schools. Statistics prove suspended students are less likely to graduate, possibly because they miss the instructional time needed to proceed academically (Augustine, et al., 2018). Restorative programs have gained momentum in the educational community as an approach to reduce suspension rates, help improve absenteeism and school climate. If educational institutions can improve relationships with the students and staff and build a sense of community within the school, it might bring results of students less inclined to misbehave and have a desire to attend school.

A study conducted in the Pittsburg Public School district in 2015-16 and 2016-17 represented one of the first controlled trials of the effects of restorative practices on school climate and suspension rates (Augustine, et al., 2018). The researchers found that restorative justice community school-wide programs have several positive effects. An improvement in overall school climates through the perception of the teachers, a decrease in overall suspension rates, and a reduction in the suspension rates between low and higher income students.

One of the most comprehensive evaluations of restorative practices used in schools was conducted in the United Kingdom. This study employed surveys and interviews with over 5,000 students, 1,150 educators, and 600 community members (Bitel, 2005). Many schools that had established restorative practice programs saw a reduction in bullying, improved over-all school

behavior, and educators who looked for different ways to punish misbehaviors. Educators who engaged in restorative practice programs felt the school climate had improved and students were more satisfied and felt they had been dealt with more fairly in the process.

Another study of 18 schools in Scotland produced comparable accomplishments to those demonstrated in the United Kingdom. The study was conducted for two years through surveys, facilitated interviews, and focus groups where educators employed restorative practice programs (McCluskley et al., 2008). This research indicated that students felt more positive about their school experience. Attendance and staff morale improved while disciplinary referrals and rates of expulsions decreased (McCluskley et al., 2008).

A study conducted at City Springs Elementary and Middle School in Baltimore showed restorative practices brought about noticeable changes in student behavior and school climate. Many teachers at City Springs felt the restorative practice approach was just as disruptive at the beginning of the program as their student misbehaviors. Most of their attitudes from other years of experience were to suspend the kid or get them out of their classroom. When the restorative practice program was initiated and as the teachers learned to build relationships with their students by gathering them together and asking and answering questions, and the students began to share their feelings, express what's on their mind and learn about each other the teachers started to see positive results. The school started a school-wide approach of circles to defuse a disciplinary situation. The teachers were taught to quickly assemble a circle and get students to talk about what they think when they behave a certain way and ask them how they can make the situation correct. If circumstances necessitated it, the circle approach was used more formally. In these discussions, everyone should talk through an incident with parents and supporters for both students (Shah, 2012). Over two years, the school found that suspension

rates had declined in middle school. It was also concluded that a more favorable level of student engagement and school involvement was noted.

In 2015, the Lakehead School District explored the implementation of restorative practices within three of their schools. While the initial evaluation revealed positive findings, such as a decreased number of suspensions and enhanced student attitudes toward learning, it also concluded that the payback of restorative practices could be maximized when they are implemented school-wide (Lewington, 2016).

Summary

A secure, respectful, and well-ordered learning environment where teachers, administrators, students, parents, and community are engaged in social responsibility is essential to the student learning process. A progressive disciple approach is a foundation for a school approach that uses a gamut of prevention and intervention programs that addresses inappropriate student behavior and builds strategies to promote positive behaviors. Restorative practices are replacing zero-tolerance policies regarding disciplinary and school climate issues. Restorative practice programs are used as a cohesive and proactive approach where the major goal is to reintegrate students into a successful learning process (Zheng & DeJesus, 2018). A constructive learning and teaching climate is necessary for student achievement and welfare.

Research has indicated a relationship between students' success and school climate (Safe Schools Action Team, 2008). Bringing about a positive school climate is not easy and requires a whole-school line of attack. Despite the increasing popularity of restorative programs in U.S. schools, there is at this point little research on the impact of such practices in school settings (Anfara et al., 2003).

Recent studies indicated the proportion of suspensions where restorative practices were

implemented were not conclusive in reducing the overall total; however, restorative practice programs did have a positive impact on reducing the possibility of students being re-suspended. The Toronto District School Board in Canada reported almost 73% of students who participated in a restorative practice program after their first suspension did not have any suspensions in the rest of their school career; 6% more than students who had not participated in a restorative practice program at 67% (Zheng & DeJesus, 2018). While there was almost no difference in absenteeism rates in the school year when students were first suspended, the following school years, a more substantial 7% of students who had participated in a restorative practice program had a lower absentee rate than students who had not participated in a restorative practice program.

The research on restorative practice programs in schools is still developing and several of the case studies have indicated promising results with school climate, student behavior, and relationships between students and staff (Ashley & Burke, 2009). Some researchers suggest a shift in attitudes toward punishment may take one to three years, and a change to a school restorative practice program where a significant difference in school climate is documented might take up to three to five years (Evans & Lester, 2013), (Karp & Breslin, 2001).

Although a growing body of research has established that restorative practices can foster positive relationships and improved school climate, the impact of restorative practices as an intervention for students facing exclusionary discipline is still to be determined. The purpose of this study is to evaluate and provide an overview of discipline practices, discipline referrals, absenteeism, and teacher's perception about a restorative practice program initiated in a low-income middle school in east Tennessee as well as results from the use of the restorative practice has in the overall school environment.

CHAPTER 3

Research Design and Methods

Introduction

In the fall of 2019, a school-based restorative practice program was implemented at a middle school in rural East Tennessee. This program focused on improving overall discipline, attendance, and school culture. All staff members including administration, teachers, and counselors met with students regularly and participated in restorative practice circles. These circles focus on identifying the problem and developing solutions to assist students in gaining restorative and coping mechanisms. Students were able to discuss and identify valuable reasons behind their actions and discuss other ways to better address issues. The restorative practice method focused on improving the school community and culture throughout the grade levels. Through the restorative practice program and circle time, students were able to feel a sense of community and be able to identify with their peers on a deeper and personal level.

The purpose of the exploratory sequential mixed-methods study was to explore the implementation of restorative practices based on the perceptions of the teachers and overall discipline in coordination with attendance data from fall 2018 and fall 2019 in at a high school in rural East Tennessee. Data were analyzed to determine if ISS, OSS, and alternative school placement demonstrated an increase or decrease at the school level between the seventh and eighth grade students in fall 2018 and fall 2019. A focus group consisting of three seventh grade teachers, three eighth grade teachers, and one related arts teacher was interviewed to determine their perspectives regarding the implementation of the restorative practice program and the effects of overall student behavior, achievements, and school climate.

This chapter provides a rationale for the exploratory sequential mixed-methods design of the study. The populations, participants, and samples used in this study are described. Data collection methods are included along with an analysis of the data sets. This chapter will concludes with a summary of the results displayed in Chapter 4.

Rationale for Using Mixed-Methods

An exploratory sequential mixed-methods were used for this research design.

A mixed methods research represents research that involves collecting, analyzing, and interpreting quantitative and qualitative data in a single study or in a series of studies that investigates the same underlying phenomenon (Onwuegbuzie, 2008). By utilizing a mixed-methods approach with restorative practices, the study provides a more comprehensive understanding of the program and demonstrates an overall perspective of how the design added to the school-wide culture. Hughes (2016) states, "quantitative analyses employ descriptive and inferential statistics, whereas qualitative analyses produce expressive data that provide descriptive details (often in narrative form) to examine the study's research objectives. This design allows for qualitative and quantitative components to intertwine and provides a better insight into the overall school results. By beginning this with a qualitative inquiry, the teacher's perspective of school climate was identified through interviews then the quantitative instruments were used to document measurable changes.

Qualitative Components

Qualitative research methods were used to gain insight and collect valuable information regarding the teacher's perspectives on discipline and attendance in middle school. All middle school teachers were trained and given the appropriate tools to implement the restorative practice program. Focus group interviews were conducted consisting of seven teachers prior to

implementing the restorative practice program and after it was effective for a semester. The main purpose of the teacher interviews were to compare and contrast the overall school's discipline, attendance, and school climate. The responses were analyzed and used as a guide to determine the effect of restorative practices within the middle school environment.

Quantitative Components

PowerSchool reports were generated to determine student's ISS, OSS, alternative school placement, and discipline referrals in fall 2018 and fall 2019. Attendance data were generated to identify the number of days absent for each student in fall 2018 and fall 2019. Data were entered in SPSS and calculated to determine any significant differences in those components between fall 2018 and fall 2019.

Qualitative Research Questions

The following three research questions were used to address the qualitative research in the study:

- 1. What are the perceptions of effectiveness, fairness, and training/satisfaction of teachers after training (but before institution) in restorative practices?
- 2. What are the perceptions of effectiveness, fairness, and training/satisfaction of teachers after a semester of instituting restorative practices?
- 3. Have teachers' perceptions of effectiveness, fairness, and satisfaction changed regarding restorative practices in this time frame?

Quantitative Research Questions

The following seven research questions were used to address the quantitative strand in the study:

1. Is there a significant difference in the number of student discipline referrals during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

Null Hypothesis: There is no significant difference in the number of student discipline referrals during the fall 2018 semester versus the fall 2019 semester, following the institution of restorative practices?

2. Is there a significant difference in the number of out-of-school suspensions during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

Null Hypothesis: There is no significant difference in the number of out-of-school suspensions during fall 2018 semester versus the fall 2019 semester, following the institution of restorative practices?

3. Is there a significant difference in the number of in-school suspensions during fall 2018 semester versus fall 2019 semester, following the institution of restorative discipline practices?

Null Hypothesis: There is no significant difference in the number of in-school suspensions during the fall 2018 versus the fall 2019 semester, following the institution of restorative practices?

4. Is there a significant difference in the number of students suspended (both in-school and out-of-school) during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

Null Hypothesis: There is no significant difference in the number of students suspended (both in-school and out-of-school) during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

5. Is there a significant difference in the number of students placed in an alternative school during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

Null Hypothesis: There is no significant difference in the number of students placed in an alternative school during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

6. Is there a significant difference in the number of absences during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

Null Hypothesis: There is no significant difference in the number of absences during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

7. Is there a significant difference in the number of students who are considered "chronically absent" during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

Null Hypothesis: There is no significant difference in the number of students who are considered "chronically absent" during fall 2018 semester versus the fall 2019 semester, following the institution of restorative practices?

Population

The population for this study was generated from a middle school in rural East

Tennessee. In 2018-2019 there were a total of 98 seventh grade students and a total of 99 eighth grade students. There were 49 males and 49 females in seventh grade and 55 males and 44 females in eighth grade. The student population was 92.9% Caucasian, 5% African American, and the remaining 2.1% Hispanic or Latino. In 2019-2020 there were a total of 103 seventh grade students and a total of 112 eighth grade students. There were 49 males and 54 females in seventh grade and 66 males and 46 females in eighth grade. The student population for seventh and eighth grade was 92.5% Caucasian, 3.3% African American, and the remaining 4.2% Hispanic or Latino.

Sample

To obtain a sample, the population were generated for seventh and eighth grade students from PowerSchool. The total student size was 161 students in seventh and eighth grade. From this list student information was identified such as attendance, ISS, OSS, alternative school, and the number of discipline referrals. The sample size included all students who were enrolled in the East Tennessee middle school in fall 2018 and fall 2019. Seven teachers across grade levels were identified by the administration to participate in a focus group to obtain information regarding teacher's perspective on the restorative practice program before and after implementation.

Instrumentation

Qualitative Instrumentation

Qualitative data were collected in the form of focus group interviews. A focus group is a small group of people who are knowledgeable and can offer valuable information about the subject matter. "Focus groups are used for generating information on collective views, and the

meanings that lie behind those views. They are also useful in generating a rich understanding of participants' experiences and beliefs" (Gill, Stewart, Treasure, Chadwick, 2008, p. 204). Focus group interviews were used to gain insight into the teacher's perceptions of attendance, discipline, and overall school climate. Focus group interviews are completed and facilitated in a small group setting. The focus group questions are guided to generate an overall idea of the school climate and culture pertaining to attendance and discipline. The focus group met based on administrative decisions pertaining to which teachers would be participating in the study beginning in fall 2019 and then again at the conclusion of fall 2019 to ensure triangulation of qualitative data. The questions, which were asked in the focus group interviews, were asked in a general setting with the seven teachers providing input among all grade levels as a whole. Focus group interviews were held in the library and facilitated by the researcher. The qualitative data were analyzed using the Creswell's six generic approaches to analyzing qualitative data.

Quantitative Instrumentation

The quantitative data were gathered through PowerSchool generated reports and the school's discipline referral forms. The reports included fall 2018 and fall 2019 attendance, ISS, OSS, and alternative school placement information. For the quantitative strand only students enrolled in the seventh and eighth grades during fall 2018 and fall 2019 were analyzed. Students were given a number and compared based on if they were enrolled in the middle school throughout fall 2018 and fall 2019. Information was entered in the SSPS program and tests were generated to compare the data.

Data Collection and Procedures

Before the study was conducted, a proposal was sent to Milligan College IRB for approval. Consent to conduct research was granted by the school system's superintendent and the

principal of the middle school. A consent form was given to the teachers involved in the focus group regarding the research. A copy of the consent form was provided to every teacher. Once informed consent was received, the restorative practice program was implemented. The researcher kept all consent forms.

Participants experienced minimal risk during the study. Participants could experience some level of anxiety from being questioned about their classroom and school environment and when asked to expand on any negative aspects generated utilizing the restorative practice program. All information was provided to the teachers in regard to how the information that is collected was utilized for the study.

Focus group interviews were conducted by contacting seven teachers. Three teachers from seventh grade, three teachers from eight grades, and one related arts teacher were selected to participate in the focus group interviews to gain insight into the teacher's perspectives on the implementation of the restorative practice program. The researcher asked scripted questions and teachers were recorded with sound only via an iPad. The teachers freely discussed and answered the generated questions within the small group setting. Two sessions were conducted one in August 2019 and the other in December 2019 following one semester of implementing the restorative practice program.

Questions asked during the interviews were related to the perceptions based on attendance, discipline, and overall school climate. The qualitative data were scored using Creswell's six generic steps (Creswell, 2009). Information from the focus group was used to gain insight into teacher's perspectives after implementing the restorative practice program pertaining to attendance, discipline, and overall school climate.

In August 2019, PowerSchool data were accessed to show fall 2018 discipline records

and discipline referrals were provided to the researcher. The quantitative data were collected again in December 2019 to reflect the data collected for the fall 2019 semester. The quantitative data were inputted in SSPS and statistical tests were run to compare fall 2018 and fall 2019 data.

Copies of all information were kept in a file along with hard copies of the teacher's consent forms and transcription information. All information was kept in safe storage to protect the confidentiality and integrity of the research project.

Although the researcher participated in conducting the interviews, it was the role of the teachers to implement the restorative practice program in the classroom. The researcher did not implement the restorative practice circles or methods into the school or the classroom. The researcher provided the administration with the results from the focus group interview questions in August 2019 and December 2019 and all quantitative results that were generated on discipline and attendance.

Data Analysis

Qualitative Strand

A multiple step qualitative data analysis utilizing Creswell's six generic steps approach was used and followed after qualitative data were collected (Creswell, 2009). The following steps were used to analyze the qualitative date:

Organized and prepared the data for the analysis: the researcher transcribed the focus
interview questions from the audio recording and typed it on a document for further
analysis and use.

- 2. The researcher read and interpreted the data for a general understanding of the topic and teacher perception: responses were noted and categorized by teacher responses to better understand the dynamics and responses in order.
- Detailed analysis with a coding process: teacher responses were organized into categories. Similar responses were grouped to assist in developing themes and similar thoughts.
- 4. Coded responses to generate theme: each category was analyzed and grouped with other categories to develop a related theme or response.
- 5. Represented themes in the qualitative narrative: A flow chart was created to display teacher responses for each theme. Phrases and keyword that were taken from teacher responses were used to ensure understanding of the material.
- 6. Interpreted the meaning of the data: the researcher analyzed data to determine the perceived need for the restorative practice program and changes to the overall school climate.

The qualitative steps were taken to increase and improve the validity and reliability of the analysis of the data (Creswell, 2009). By utilizing these steps, the researcher was able to organize the data, develop a conclusion about teacher perspectives pertaining to the overall school climate. Based on the analysis of results, the researcher was able to develop an assumption of the school environment before and after utilizing the restorative practice program.

Quantitative Strand

Data from PowerSchool pertaining to discipline and attendance was entered into SSPS for statistical analysis. For quantitative question #1, a paired t-test was used to test for a significant difference in the number of student discipline referrals during fall 2018 and fall 2019 semester, following the institution of the restorative practice program. For quantitative question #2, a paired t-test was used to test for a significant difference in the number of out-of-school suspensions during fall 2018 and fall 2019 semester, following the institution of restorative practice program. For quantitative question #3, a paired t-test was used to test for a significant difference in the number of in-school suspensions during fall 2018 and fall 2019 semester, following the institution of restrictive practices. For quantitative question #4, a paired t-test was used to test for a significant difference in the number of students suspended both in-school and out-of-school during the fall 2018 and fall 2019 semester, following the institution of restorative practices. For quantitate question #5, a paired t-test was used to test if the number of students placed in alternative school during fall 2018 and fall 2019, following the institution of restorative practices. For quantitative question #6, a paired t-test was used to test for a significant difference in the number of students absent during the fall 2018 and fall 2019 semester, following the institution of restorative practices. For quantitative question #7, a paired t-test was used to test for a significant difference in the number of chronically absent students during the fall 2018 and fall 2019, following the institution of restorative practices.

Summary

The exploratory sequential mixed-methods design encompassed both qualitative and quantitative methods to prove a more comprehensive understanding of the data. By conducting focus group interviews, the researcher has access to individual teacher experiences and their perspectives regarding the overall school climate. After analyzing the qualitative data, a better understanding of classroom teacher's perspectives were gained and acknowledge on how they felt the restorative practice program contributed to the school climate and overall discipline and attendance. ISS, OSS, alternative school placement, and attendance were access through PowerSchool for fall 2018 and fall 2019 semester. Discipline referrals were given to the researcher to enter in SSPS along with all the other viable data gathered. Finally, a comprehensive analysis was completed to show the overall comparison of implementing the restorative practice program fall 2018 versus fall 2019.

CHAPTER 4

Data Analysis and Findings

Introduction

The purpose of this sequential exploratory mixed-methods study was to explore the effects of restorative practices on ISS, OSS, an alternative school placement, attendance, and teacher perceptions. The population studied was seventh and eighth grade students in a Title I middle school in rural East Tennessee.

Qualitative data assessing teacher's perceptions of the overall school climate before and after the implementation of restorative practices were collected in the form of focus group interviews in August 2019 and December 2019. The sample consisted of three seventh grade teachers, three eighth grade teachers, and one related arts teacher. The teachers who participated in the focus group were asked questions related to the use and implementation of the restorative practice program. The following research questions guided the interview:

- 1. What are the perceptions of effectiveness, fairness, and training/satisfaction of teachers after training (but before institution) in restorative practices?
- This research question addresses interview questions 1a, 2a, 3a, 4a, 5a, 6a, 7a, 8a, 9a, 10a, 11a, and 12a.
- 2. What are the perceptions of effectiveness, fairness, and training/satisfaction of teachers after a semester of instituting restorative practices?

This research question addresses interview question 1b, 2b, 3b, 4b, 7b, 8b, 9b, and 11b.

3. Have teachers' perceptions of effectiveness, fairness, and satisfaction changed regarding restorative discipline practices in this time frame?

This research question addresses interview questions 5b, 6b, 10b, and 12b.

Focus group questions were generated to address the qualitative research questions and gain insight into the use of restorative practices before and after implementation. These questions were asked to the focus group participants during the August 2019 and December 2019 interviews. The researcher used the data from the interviews to sort, code, and identify themes.

Quantitative data were collected for each student enrolled in fall 2018 and fall 2019 regarding ISS, OSS, alternative school placement, and attendance were collected through PowerSchool. Discipline referral forms for each student were counted and all data were inputted into SSPS. The following quantitative research questions were addressed:

- 1. Is there a significant difference in the number of student discipline referrals during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?
- 2. Is there a significant difference in the number of out-of-school suspensions during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?
- 3. Is there a significant difference in the number of in-school-suspensions during the fall 2018 semester versus fall 2019 semester, following the restorative practices?
- 4. Is there a significant difference in the number of suspensions (both in-school and out-of-school) during the fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?
- 5. Is there a significant difference in the number of students placed in an alternative school during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

- 6. Is there a significant difference in the number of absences during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?
- 7. Is there a significant difference in the number of students who are considered chronically absent during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

In this chapter, sample and population demographics are identified and described for each part of the study. Qualitative data are shown in table form with themes listed based on the responses. Keywords and phrases from the teacher's responses are included and an asterisk indicates keywords or phrases that were repeated by teachers in the interviews. Quantitative data are shown in the form of tables to identify the levels of significance for each research question.

Demographic Data

The population studied was all seventh and eighth grades students that were enrolled in fall 2018 and fall 2019 in a rural Title I middle school in East Tennessee. Three teachers that taught seventh and eighth grade along with one related arts teacher was interviewed. These interviews were conducted as focus groups. Each question was read and the teachers had appropriate time to respond. These responses were coded by hand by the researcher.

Qualitative Strand Findings

Responses to Interview Question 1a (August 2019): Thinking back over this past school year, tell me some ways the school environment was conducive to learning.

Common themes emerged such as communication and cohesiveness. Teacher responses are presented in Table 1a.

Table 1a. Teacher Responses to Interview Questions 1

Themes	Number of Responses	Key Words, Phrases, & Quotes
Communication	4	- Classroom Expectations*
		- Procedures
	- "We cor	mmunicated classroom expectations."
Cohesive	3 - "All stude	- Working together - "We are a team" nts and teachers work well together."

^{*} Indicates a response that was given by multiple participants

Responses to Interview Question 1b (December 2019): Thinking back over the fall semester, how has the school environment been more conducive to learning after implementing restorative practices. Common themes emerged such as communication, relationships, and organization. Teacher responses are presented in Table 1b.

Table 1b. Teacher Responses to Interview Questions 1b

Thomas	Number of Degranges	Vay Words Phrasas and Ouatas
Themes Communication	Number of Responses	Key Words, Phrases, and Quotes
Communication	5	- Expectations*
		- Consequences for actions
		 Conversations and circles
	 "Conversations wi 	th other teachers that focus on leadership."
	- "Communication v	vith other teachers discussing issues."
Relationships	3 - Ot	her teachers, students, and administration
read on the second of the seco		ationships have been built with everyone."
Organization	2	- Overall organization
		- Higher Standards
- "We are trying to	o hold them to a higher star	ndard both academically and behaviorally."

^{*} Indicates a response that was given by multiple participants

Responses to Interview Question 2a (August 2019): Thinking back over this past school year, give me examples of the main issues regarding discipline and attendance in this school climate. Common themes emerged such as tardy and attendance. Teacher responses are presented in Table 2a.

 Table 2a. Teacher Responses to Interview Questions 2a

Themes	Number of Responses	Key Words, Phrases, & Quotes
Tardy	3	- Late to class
		- Too many tardies
		Spending time at lockers*
	- "Studen	ts are lingering too long at the lockers."
	- "Studen	ts stay too long in the bathrooms."
Attendance	4	- Struggling at home
		- Missing class instruction
		- Same students every year
	- "Students who struggle at ho	me have a hard time getting to school."

^{*} Indicates a response that was given by multiple participants

Responses to Interview Question 2b (December 2019): Thinking back over the fall semester, name issues regarding discipline and attendance which have improved after implementing restorative practices. Common themes emerged such as communication and follow through. Teacher responses are presented in Table 2b.

Table 2b. Teacher Responses to Interview Questions 2b

Themes	Number of Responses	Key Words, Phrases, & Quotes
Communication	5	- Talking to parents
		- Conversations with students
	- "We are involving parents	and students in attendance meeting."
Follow through	2	- Tardy policies enforced
C		- Attendance policies enforced
	- "Policies are in place	ce and everyone is following through."

Responses to Interview Question 3a (August 2019): Describe how you feel a restorative practice program will have an effect on the school climate and teacher satisfaction. Common themes emerged such as communication and leadership. Teacher responses are presented in Table 3a.

Table 3a. Teacher Responses to Interview Questions 3a

Themes	Number of Responses	Key Words, Phrases, and Quotes
Communication	- Expe - "We comm	ents having conversations and a voice* ectations unicate our classroom expectations." have a voice in making decisions."
Leadership		ent responsibility for actions to take responsibility for their actions."

^{*} Indicates a response that was given by multiple participants

Responses to Interview Question 3b (December 2019): Tell me some ways the school climate and teacher perceptions have improved upon the implementation of restorative practices.

Common themes emerged such as meaningful relationships and comfortability. Teacher responses are presented in Table 3b.

Table 3b. Teacher Responses to Interview Questions 3b

Themes	Number of Responses	Key Words, Phases, & Quotes
Meaningful Relationships	3 - "The office is 1	 Non-judgmental Communication with teachers Student and teacher connection becoming to them a non-judgmental area."
Comfortability		- Administration, teachers, and students as are more open to talk to administration." able to better communicate with teachers."

Responses to Interview Question 4a (August 2019): Describe how you feel a restorative practice program will have an effect on discipline and attendance in your school.

Common themes emerged such as positive thinking and school climate. Teacher responses are presented in Table 4a.

Table 4a. Teacher Responses to Interview Questions 4a

Themes	Number of Responses	Key Words, Phrases, Quotes
Positive Thinking	- Relat - Grov	le Time tionships with stakeholders with mindset* n a growth mindset and think positive."
School Climate	- Studei - "Everyone is worki	one working together hts learning respect and responsibility ing together for the same goal." ng to learn respect and responsibility."

^{*} Indicates a response that was given by multiple participants

Responses to Interview Question 4b (December 2019): Give me an example of how restorative practices have improved discipline and attendance in your school. Common themes emerged such as responsibility and awareness. Teacher responses are presented in Table 4b.

Table 4b. *Teacher Responses to Interview Questions 4b*

Themes	Number of Responses	Key Words, Phrases, & Quotes
Responsibility	- Teache - Studer - "I see our stu	ts taking ownership* ers taking ownership* hts addressing issues with others heldents taking ownership for their actions." ddress issues with each other positively."
Awareness - "The students are	- Non-juo	ts becoming aware of their actions dgmental attitudes among all stakeholders nd developing non-judgmental attitudes."

^{*} Indicates a response that was given by multiple participants

Responses to Interview Question 5a (August 2019): Tell me some ways the majority of the staff believe the restorative practice can improve student behavior. Common themes emerged such as resistant to change and examples. Teacher responses are presented in Table 5a.

Table 5a. Teacher Responses to Interview Questions 5a

Table 5b. Teacher Responses to Interview Questions 5b

Themes Number of Responses Key Words, Phrases, & Quotes

Resistant to Change 2 - Thinking it is another program
- Frustrated need more time (short period)
- "Teachers are resistance to change and will think this is another program."

Examples 4 - Provide videos and proof
- Show positive outcomes
- "Teachers would like to see documented proof that showed positive outcomes."

Responses to Interview Question 5b (December 2019): What aspects of your experience with restorative practices lead you to believe the majority of the staff now believes restorative practices have improved overall student behavior? Common themes emerged such as positive conversations and student's actions. Teacher responses are presented in Table 5b.

Themes Number of Responses Key Words, Phrases, & Quotes

Positive Conversations 3 - Colleague conversations about students
- "Conversations are occurring that are essential in understanding the issues at hand."

Student's Actions 5 - Student discipline is decreasing
- Positive student conversations
- "We see more positive student conversations happening and discipline decreasing."

Responses to Interview Question 6a (August 2019: Describe how the majority of the staff believe restorative practice can improve student attendance. Common themes emerged such as building relationships and staying informed. Teacher responses are presented in Table 6a.

Table 6a. Teacher Responses to Interview Questions 6a

Themes	Number of Responses	Key Words, Phrases, & Quotes
Building Relationships	- - "Everyone in the building d	Demonstrating care Demonstrating love Demonstrating safety* emonstrates care and love for one another." s essential when developing relationships."
Staying Informed	-	Continue to improve attendance Learn effective ways to help with students* med and continue to learn ways to improve."

^{*} Indicates a response that was given by multiple participants

Responses to Interview Question 6b (December 2019: What about your experience leads you to believe the majority of the staff now believes restorative practices has improved overall student attendance. Common themes emerged such as communication and attendance policies. Teacher responses are presented in Table 6b.

Table 6b. Teacher Responses to Interview Questions 6b

Themes	Number of Reponses	Key Words, Phrases, & Quotes
Communication	4 We are staying in contact with teach	- Students talking to teachers - Contact with parents chers and parents during this process."
Attendance Policies	4 - "Students are being p	- Students in class more often - Students following policies present in class and following policies."

Responses to Interview Question 7a (August 2019): Tell me some ways the staff is familiar with the purpose and methods of restorative practices. Common themes emerged such as trainings and discussions. Teacher responses are presented in Table 7a.

Table 7a. Teacher Responses to Interview Questions 7a

Themes	Number of Reponses	Key Words, Phrases, & Quotes
Trainings	3	- Presentations
		- PowerPoints
	- "We need trainings a	nd presentations to implement this properly."
Discussions	5	- Conversations with colleagues
		- Conversations with administration
	- "Important conversations are or	ecurring with colleagues and administration."

Responses to Interview Question 7b (December 2019): Tell me some ways the staff is more comfortable with the purpose and methods of restorative practices. Common themes emerged such as benefits and comfortability. Teacher responses are presented in Table 7a. All seven teachers were interviews.

Table 7b. Teacher Responses to Interview Questions 7b

Themes	Number of Responses	Key Words, Phrases, & Quotes
Benefits	4	- Documented results
		- Student's attitudes
- "We see	documented results of success and st	udent's attitudes are improving daily."
Comfortability	3	 Provided resources to assist
		- Continued training
	- "Teachers are feeling more co	omfortable with the trainings offered."

Responses to Interview Question 8a (August 2019): Describe how the staff feels about implementing restorative practices with the majority of the students of Sullivan North Middle.

Common themes emerged such as trainings and discussions. Teacher responses are presented in Table 8a.

Table 8a. Teacher Responses to Interview Questions 8a

Themes	Number of Reponses	Key Words, Phrases, & Quotes
Hopeful	2	- Feeling excited
		- Anxious for this to work
		- Want the best for the students
		- Willing to do whatever for the students
	- "We are feeling excited	d and will do what we can for the students."
	- "We want the best for o	our students."
Nervous	2	 Hoping it is beneficial Hoping to understand the program
	- "We are hopeful this program	will be beneficial."
	- "I hope to fully understand the	e program and how it will help our students."

Responses to Interview Question 8b (December 2019): Describe how the staff feels about implementing restorative practices with the majority of the students. Common themes emerged such as beneficial and results oriented. Teacher responses are presented in Table 8b.

Table 8a. Teacher Responses to Interview Questions 8b

Themes	Number of Responses	Key Words, Phrases, & Quotes
Beneficial	4	- Excited about the changes
		 Students seem happy
		 Climate is encouraging
	- "O	ur students seem happy."
	- "E	veryone is excited about the changes."
Results Oriented	4	 Results are positive
		- Staff is optimistic of future results
	-	"We are seeing positive results."
	-	"We look for future results to positive."

Responses to Interview Question 9a (August 2019): Describe how you feel the school climate will improve with the result of restorative practices. Common themes emerged such as cohesiveness and mindset. Teacher responses are presented in Table 9a.

Table 9a. Teacher Responses to Interview Questions 9a

Themes Cohesiveness	Number of Responses 3	Key Words, Phrases, & Quotes - Everyone on the same page
	- "Everyone is on the san	- Families working with the school ne page with circles including families."
Mindset	5	- Growth mindset* - Positive Outcomes "We must have a mindset for growth."

^{*} Indicates a response that was given by multiple participants

Responses to Interview Question 9b (December 2019): Tell me some ways the school climate has improved with the result of restorative practices. Common themes

emerged such as root cause analysis and positive environment. Teacher responses are presented in Table 9b.

Table 9b. Teacher Responses to Interview Questions 9b

Themes	Number of Responses	Key Words, Phrases, & Quotes
Root Cause Analysis	4 - "Teachers are beginning	 Understanding the main issue Gaining insight to helpful resources ing to understand the root cause of issues."
Positive Environment	6	 More positive thinking* More positive attitudes
- "Teachers seen		ore positive thinking about the program." out implantation of restorative practices."

^{*} Indicates a response that was given by multiple participants

Responses to Interview Question 10a (August 2019): Give me some examples of how the staff believes restorative practices will improve the overall school culture. Common themes emerged such as better attitudes and better relationships. Teacher responses are presented in Table 10a.

Table 10a. Teacher Responses to Interview Questions 10a

Themes	Number of Responses	Key Words, Phrases, & Quotes
Better attitudes 6 -		ve work environment
	- Stude	nts happier
	- Better	r overall behavior
- "We ho	pe this program will provide bet	ter behavior and a happier environment."
Better Relationships	7 - All sta	keholder included in decisions
	- All stal	keholders build meaningful relationships
- "All staff nee	ds to build relationships with oth	ner staff members, students, and parents.

Responses to Interview Question 10b (December 2019): Give me some examples of how the staff believes restorative practices has improved the overall school culture. Common themes emerged such as consistency and positive thinking. Teacher responses are presented in Table 10b.

Table 10b. Teacher Responses to Interview Questions 10b

Themes	Number of Responses	Key Words, Phrases, & Quotes
Consistency	 4 - All stakeholders are consistent with issues - Discipline issues are consistent - "Everyone needs to be on the same page with school issues. 	
Positive Thinking	- Better attitudes ar	tive with the implementation and thinking about school related issues." as a positive attitude with the program."

Responses to Interview Question 11a (August 2019): In your experience what leads you to believe that students will handle conflict with other students as a result of restorative practices. Common themes emerged such as positive and conflict resolution. Teacher responses are presented in Table 11a.

Table 11a. Teacher Responses to Interview Questions 11a

Themes	Number of Responses	Key Words, Phrases, and Quotes
Positive	5	- Better outlook on behavior and attitudes
		- Better relationships with peers
	- "I hope this will provide a be	etter outlook and relationships with peers."
Conflict Resolution	3 - "I am glad that this program	- Working out issues between peers -Working together with all stakeholders will help bring all stakeholders together."

Responses to Interview Question 11b (December 2019: Describe how you feel the students can handle conflict with other students as a result of restorative practices. Common themes emerged such as student voice and collaboration. Teacher responses are presented in Table 11b.

Table 11b. Teacher Responses to Interview Questions 11b

Themes	Number of Responses	Key Words, Phrases, & Quotes
Student Voice	5 "Students are discussing issues wit	 Students advocating for themselves Discussing issues with peers* Discussing issues with teachers Students being able to discuss issues th other students and teachers more often."
Collaboration - "I lik	4 ke that everyone is working togethe	 Everyone working together Talking through issues Teachers working together Students talking with each other r and dealing with issues collaboratively."

^{*} Indicates a response that was given by multiple participants

Responses to Interview Question 12a (August 2019): In your experience what leads you to believe that the staff will handle conflict with students as a result of restorative practices. Common themes emerged such as belief in the program and respectful. Teacher responses are presented in Table 12a.

Table 12a. Teacher Responses to Interview Questions 12a

Themes	Number of Responses	Key Words, Phrases, & Quotes
Belief in the program	4 - Thinking positivity about the results - Working together as a school to receive the best results - "We have belief in the program." - "We all need to work together to gain overall positive results."	
Respectful	4	Demonstrate concernRespect other's viewpoints
- "I	am glad that everyone is respect	tful of other's viewpoints and opinions."

Responses to Interview Question 12b (December 2019): Describe how the staff feels they can handle conflict with students as a result of restorative practices. Common themes emerged such as belief in the program and respectful. Teacher responses are presented in Table 12b.

Table 12b. Teacher Responses to Interview Questions 12b

Themes	Nı	umber of Responses	Key Words, Phrases, & Quotes
Trust		5	- Builds trust within the school
			- Honesty
			 Lasting relationships
			sting relationships and has built trust." ealing with difficult tasks."
Positive viewp	oints	5	Respectful conversationHopeful"We have maintained hope and respect."

The themes in the qualitative data revealed that change of teacher's perceptions regarding a restorative practice implementation provided meaningful relationships with all stakeholders and a more positive school climate. The data from SSPS were used to compute the quantitative data to determine and significant changes in ISS, OSS, alternative school placement, and absenteeism.

Quantitative Strand Findings

Research Question 1. Is there a significant difference in the number of student discipline referrals during the fall 2018 semester versus the fall 2019 semester, following the institution of restorative practices?

Research Hypothesis: There is a significant difference in the mean number of discipline referrals per student issued before and after the institution of restorative discipline practices.

Null Hypothesis: There is no significant difference in the mean number of discipline referrals issued per student before and after the institution of restorative discipline practices.

A paired sample t-test was conducted to determine if there was a significant difference between the mean number of referrals issued per student in 2018 and 2019 after the restorative practices had been implemented. The results did not show a significant difference (t(159)= .768, p= .444) between the mean number of referrals per student in 2018 (M = .88, sd = 2.032) and the mean number of referrals per student in 2019 (M = .78, sd = 1.488). The results of the paired samples t-test are displayed in Table 1.

Table 1. Paired Samples t-Test Comparing Fall 2018 and Fall 2019 Discipline Referrals

<u>Variable</u>	M	SD	df	t	<u>p</u>
Discipline Referrals 2018 Discipline Referrals 2019	.88 .78	2.032 1.488	159	.768	.444

^{*}*p* < .05

Research Question 2. Is there a significant difference in the number of out-of-school suspensions during the fall 2018 semester versus the fall 2019 semester, following the institution of restorative practices?

Research Hypothesis: There is a significant difference in the mean number of out-of-school suspensions per student issued before and after the institution of restorative discipline practices.

Null Hypothesis: There is no significant difference in the mean number of out-of-school suspensions per student issued before and after the institution of restorative discipline practices.

A paired sample t-test was conducted to determine if there was a significant difference between the mean number of out-of-school suspensions per student issued in 2018 and 2019 after the restorative practices had been implemented. The results did not show a significant difference (t(160)=-1.36, p=.174) between the mean number of out-of-school suspensions per student in 2018 (M=.19, sd=.863) and after the restorative practices were implemented in 2019 (M=.34, sd=1.095). The results of the paired samples t-test are displayed in Table 2.

Table 2. Paired Samples t-Test Comparing Fall 2018 and Fall 2019 Out-of-School Suspensions

<u>Variable</u>	M	SD	df	t	<u>p</u>
OSS 2018	.19	.863	160	-1.36	.174
OSS 2019	.34	1.095			

^{*} *p* < .05

Research Question 3. Is there a significant difference in the number of in-school suspensions during the fall 2018 semester versus the fall 2019 semester, following the restorative practices?

Research Hypothesis: There is a significant difference in the mean number of in-school suspensions per student issued before and after the institution of restorative discipline practices.

Null Hypothesis: There is no significant difference in the mean number of in-school suspensions per student issued before and after the institution of restorative discipline practices

A paired t-test was conducted to determine if there was a significant difference between the mean number of in-school suspensions per student issued in 2018 and 2019 after the restorative practices had been implemented. The results showed a significant difference (t(160)= -3.085, p=

.002, ES= .286). The mean number in-school suspensions per student were significantly less in 2018 (M = .23, sd = .823) than after restorative practices were implemented in 2019 (M = .62, sd = 1.743). The effect size is .286, indicating the implementation of restorative practices showed a small effect on the number of in-school suspensions per student. The results of the paired samples t-test are displayed in Table 3.

Table 3. Paired Samples t-Test Comparing Fall 2018 and Fall 2019 In-School Suspensions

<u>Variable</u>	M	SD	df	t	p	ES
ISS 2018 ISS 2019	.23 .62	.823 1.743	160	-3.085	.002*	.286

^{*}*p* < .05

Research Question 4. Is there a significant difference in the number of suspensions (both in-school and out-of-school) during the fall 2018 semester versus the fall 2019 semester, following the institution of restorative practices?

Research Hypothesis: There is a significant difference in the mean number of suspensions per student issued before and after the institution of restorative discipline practices.

Null Hypothesis: There is no significant difference in the mean number of suspensions per student issued before and after the institution of restorative discipline practices

A paired t-test was conducted to determine if there was a significant difference between the mean number of suspensions per student issued in 2018 and 2019 after the restorative practices had been implemented. The results showed a significant difference (t(160)= -3.058, p= .003, ES= .272). The mean number of in-school and out-of-school suspensions per student were significantly less in 2018 (M = .4224, sd = 1.46) than after the restorative practices were implemented in 2019 (M = .9565, sd = 2.36). The effect size is .272, indicating the implementation of restorative practices had a small effect on the number of suspensions per student. The results of the paired samples t-test are displayed in Table 4.

Table 4. Paired Samples t-Test Comparing Fall 2018 and Fall 2019 In-School and Out-of School Suspensions

<u>Variable</u>	M	SD	df	t	p	<u>ES</u>
ISS &OSS 2018 ISS & OSS 2019	.4224 .9565	1.46 2.36	160	-3.058	.003*	.272

^{*} *p* < .05

Research Questions 5. Is there a significant difference in the number of students placed in an alternative school during the fall 2018 semester versus the fall 2019 semester, following the institution of restorative practices?

Research Hypothesis: There is a significant difference in the mean number of alternative school placements per student before and after the institution of restorative discipline practices.

Null Hypothesis: There is no significant difference in the mean number of alternative school placements per student before and after the institution of restorative discipline practices.

A paired t-test was conducted to determine if there was a significant difference between the mean number of alternative school placements per student in 2018 and 2019 after the restorative practices had been implemented. The results showed a significant difference t(160)= -2.019, p= .045, ES= .181). The mean number of alternative school placements per student were significantly less in 2018 (M = .00, sd = .000) than after alternative school placements per student in 2019 (M = .02, sd = .156). The effect size is .181, indicating the implementation of restorative practices had a small effect on the number of alternative school placements per student. The results of the paired samples t-test are displayed in Table 5.

Table 5. Paired Samples t-Test Comparing Fall 2018 and Fall 2019 Alternative School Placement

<u>V</u> ariable	M	SD	df	t	р	ES
Alternative School 2018 Alternative School 2019	.00 .02	.000 .156	160	-2.019	.045*	.181

^{*} *p* < .05

Research Question 6. Is there a significant difference in the number of absences during the fall 2018 semester versus the fall 2019 semester, following the institution of restorative practices?

Research Hypothesis: There is a significant difference in the mean number of student absences before and after the institution of restorative discipline practices.

Null Hypothesis: There is no significant difference in the mean number of student absences before and after the institution of restorative discipline practices.

A paired t-test was conducted to determine if there was a significant difference between the mean number of student absences in 2018 and 2019 after the restorative practices had been implemented. The results did not show a significant difference between t(160) = -.041, p = .967) the mean number of student absences in 2018 (M = 9.10, sd = 7.326) and the mean number of student absences in 2019 (M = 9.12, sd = 6.833). The results of the paired samples t-test are displayed in Table 6.

Table 6. Paired Samples t-Test Comparing Fall 2018 and Fall 2019 Attendance

<u>Variable</u>	M	SD	df	t	<u>p</u>
Attendance 2018	9.10	7.320	160	041	.967
Attendance 2019	9.12	6.833			

^{*} *p* < .05

Research Question 7. Is there a significant difference in the number of students who are considered chronically absent during the fall 2018 semester versus the fall 2019 semester, following the institution of restorative practices?

Research Hypothesis: There is a significant difference in the mean number of chronically absent students before and after the institution of restorative discipline practices.

Null Hypothesis: There is no significant difference in the mean number of chronically absent students before and after the institution of restorative discipline practices.

A paired t-test was conducted to determine if there was a significant difference between the mean number of chronically absent students in 2018 and 2019 after the restorative practices had been implemented. The results did not show a significant difference between t(159) = -.589, p = .557) between the mean number of chronically absent students in 2018 (M = .44, sd = .498) and chronically absent students in 2019 (M = .47, sd = .501). The results of the paired samples test are displayed in Table 7.

Table 7. Paired Samples t-Test Comparing Fall 2018 and Fall 2019 Chronically Absent

<u>Variable</u>	M	SD	df	t	<u>p</u>
Chronically Absent 2018 Chronically Absent 2019	.44 .46	.498 .501	160	589	.557

^{*} *p* < .05

Summary

The quantitative data showed the mean number of in-school and out-of-school combined suspensions, alternative school, and in-school assignments were significantly less in fall 2018 to fall 2019. The mean number of student discipline referrals, absenteeism, out-of-school suspensions, and chronic absenteeism showed no significant difference from fall 2018 to fall 2019. The qualitative data showed the focus group felt perceptions on the effectiveness, fairness, and satisfaction did produce a change in the overall school climate from fall 2018 to fall 2019. The quantitative and qualitative research will be further discussed in Chapter 5.

CHAPTER 5

Summary, Conclusions, Discussion, and Recommendations

Purpose of the Study

The purpose of this exploratory mixed-methods study was to examine the effects of a restorative practice program on discipline, discipline referrals, absenteeism, and teacher perspectives in a rural Title I middle school. By allowing the focus group to use restorative practices in their educational programs, the educators were able to make changes in the discipline and absenteeism patterns within their classrooms and at the respective grade level. The restorative practice program placed emphasizes respect, building relationships, and meeting the needs of the school. The following research questions guided the study:

- 1. Is there a significant difference in the number of student discipline referrals during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?
- 2. Is there a significant difference in the number of out-of-school suspensions during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?
- 3. Is there a significant difference in the number of in-school suspensions during fall 2018 versus fall 2019 semester, following the institution of restorative practices?
- 4. Is there a significant difference in the number of students suspended (both in-school and out-of-school) during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?

- 5. Is there a significant difference in the number of students placed in an alternative school during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?
- 6. Is there a significant difference in the number of absences during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?
- 7. Is there a significant difference in the number of students who are considered "chronically absent" during fall 2018 semester versus fall 2019 semester, following the institution of restorative practices?
- 8. What are the perceptions of effectiveness, fairness, and training/satisfaction of teachers after training (but before institution) in restorative practices?
- 9. What are the perceptions of effectiveness, fairness, and training/satisfaction of teachers after a semester of instituting restorative practices?
- 10. Have teachers' perceptions of effectiveness, fairness, and satisfaction changed regarding restorative practices in this time frame?

Discipline data were obtained to provide a quantitative examination of the use of a restorative practice program. Focus group interviews were used to gain insight on the perception of effectiveness and satisfaction of the restorative practice program initiated within the respected grade level. The interviews provided a perceived level of satisfaction of training before and after the initiation of restorative practices.

This study was conducted to provide administrators and educators with data and information on several key discipline and absenteeism issues within a school setting along with providing educators key viewpoints regarding the implementation of a restorative practice program. The data collected is related to restorative practices in a low-income middle school

setting. The review of this study focuses on circles and group conferencing and the data tracked by individual students and their recidivism rate with behavioral issues. This study should provide educators with information on the implantation and effectiveness of the family group model when working with students and their families.

Summary of Findings

Quantitative Strand Findings

Analysis of the quantitative data revealed three significant changes before and after the implementation of a restorative practice program in a low-income middle school. The mean number of students who were assigned in-school and out-of-school combined suspensions was significantly less in fall 2018 than in fall 2019. The mean number of students who were assigned in-school suspension was significantly less in fall 2018 than in fall 2019. The mean number of students who were assigned alternative school was significantly less in fall 2018 than in fall 2019. The quantitative analysis also displayed no significant difference in discipline referrals, absenteeism, out-of-school suspensions, and chronic absenteeism between the fall 2018 and fall 2019 semesters.

Qualitative Strand Findings

Analysis of the qualitative data revealed that before implementation of the program, the focus group felt that restorative practices would produce a change in the school climate. After program implementation, the focus group felt the restorative practice program did produce more meaningful relationships with stakeholders. Also, the group felt students did develop comfort levels conducive to open discussions about their behavior and life choices. The interview data revealed school climate did improve after the implementation of the restorative practice program regarding attendance and behavioral concerns. The teachers reported students began to take on

leadership roles and demonstrate a sense of responsibility among their peers. The teachers felt more comfortable in handling difficult school situations and were able to develop a collaborative dialogue with other teachers to address their students' needs. These findings were used to regulate the restorative practice program within the school environment and continue to provide teachers and students with the resources to enhance the school climate.

Discussion of Findings

Quantitative Strand Findings

Research Question 1. This research question focused on the number of discipline referrals during fall 2018 versus fall of 2019. The t-test results showed no significant difference between fall 2018 and fall 2019 on the mean number of referrals per student. Restorative practice is utilized as a problem-solving approach to school discipline and the reduction of discipline referrals. Research suggests restorative practice is a proactive approach to school discipline issues that build a classroom around common violations through discussions to connect students to their classroom community which results in increasing compassion and empathy (Riley, 2018). Based on the findings, the teachers have not changed their approach for referring students to the administration. Teachers were limited due to scheduling issues that were in place before the implementation of the restorative practice program. The circle process was used in the implementation of the restorative practice program; however, due to scheduling, teachers felt constrained in implementing a continual restorative practice circle process in their daily schedule. The staff felt the need for a better schedule when implementing circles outside of the regular class time. The focus group felt they needed more time to be able to institute the circle process within the school day and this is possibly the reason the discipline referrals being sent to

administration did not change. The circle is beneficial because it assists students with developing problem-solving skills and identification of the root cause of the behavior.

Research Question 2. This research question focused on the number of out-of-school suspensions during fall 2018 versus fall 2019. The t-test concluded there was no significant difference between fall 2018 and fall 2019 on the mean number of out-of-school suspensions. The out-of-school suspensions did not change due to the absence of reoccurring restorative circles within the classroom and not providing the students with the necessary skills for a constructive behavioral arrangement during this time frame. The staff and administration felt students needed to remain in the school environment rather than in an out-of-school setting. The staff felt the elements of the restorative justice program did not reach the offender with a shared understanding of the program. Providing more skills for allowing the victim and the offender ample time for reflection and using the circle process to achieve the attended goal might be beneficial in the future.

Research Question 3. This research question focused on the number of in-school suspensions during fall 2018 versus the fall 2019. The results revealed fall 2018 suspensions were significantly less than fall 2019 suspension between the mean number of in-school suspensions. The increase in in-school suspensions could be due to the short timeframe this study examined. The students need to have time to develop alternative approaches to handle their behavior. Some researchers suggest that a shift in attitudes toward punishment may take one to three years and a shift to a school restorative practice program where a significant difference in school climate is documented might take up to three to five years (Evans & Lester, 2013), (Karp & Breslin, 2001). Research indicates discipline problems focus on insubordination and defiance of classroom instruction (Skiba, Peterson, & Williams, 1997). Restorative practices help students

develop coping mechanisms for handling behavior. When a restorative practice program is implemented, there is a focus on including the student and defiance in the school community and repairing relationships, which is done through the circle processes of victim-offender conferencing (Zehr et al, 2015). Through these processes, students can learn essential problem-solving techniques that may be converted to assist with future school incidents. While an increase was seen in the fall semester, research showed that the use of ISS provides a limited amount of change until after a three to five year institution of the restorative practice program.

Research Question 4. This research question focused on the number of suspensions both in-school and out-of-school during fall 2018 versus the fall 2019. The results revealed fall 2018 suspensions were significantly less than fall 2019 suspensions between the mean number of in-school and out-of-school suspensions. In-school suspensions likely increased in fall 2019 due to the lack of time spent developing the restorative practice program while providing the necessary tools to build relationships and to teach students how to communicate their behaviors. Out-of-school suspensions did not change significantly due to the fact school administration felt the need to keep more students in the school environment. ISS keeps students in the classroom atmosphere and school officials can punish inappropriate behavior more constructively (Haley & Tomczyk, 2000). By continuing the restorative practice program, students will learn how to repair relationships and address behavioral issues. They will become more actively engaged in their school and it will help them deal with future incidents in their lives. The ISS and OSS numbers within this research were likely increased due to the large number of ISS suspensions. Research suggests that a shift in attitude away from punishment toward a restorative practice program might take up to three years (Karp & Breslin, 2001). To create this type of attitude shift, the school administration may want to evaluate the process and utilize restorative practice

programs to understand the root cause of the discipline issues. Trainings on discipline topics need to be on-going, supported with coaching, and follow-up with the administration (Bummer, 2016). These professional trainings need to enhance not just restorative practice skills but a change in thinking about discipline, behavior, and child development (Bummer, 2016).

Research Question 5. This research question focused on the number of students placed in alternative schools during fall 2018 versus the fall 2019. The results revealed fall 2018 alternative school placements were significantly less than fall 2019 between the mean number of alternative school placements. The alternative school placements likely increased because the student rights and responsibilities handbook for the school district was revised and alternative school placements instituted where in-school or out-of-school punishments were previously administered. For example, students who were previously suspended for the use of tobacco products during fall 2018 were placed in an alternative school for this type of in the fall of 2019.

Research Question 6 & 7. These research questions focused on the difference in the number of absences and cases of chronic absenteeism during the fall 2018 semester versus the fall 2019 semester. The results indicated no significant difference in the mean number of per student or cases of chronic absenteeism. Using only one semester to evaluate attendance results does not allow adequate time for parents, students, administrators, and teachers to fully implement restorative circles on a regular basis. Teachers need more opportunities to participate in personal contact with parents regarding the restorative practice program in order to move the needle to increase attendance. Literature suggests more research needs to be conducted with interpreting the motivations for students to attend school and the understanding why students are absent or chronically absent is imperative to understanding the problem (Corvill-Smith et al., 1998). Participating in a restorative justice program and researching restorative practices are two

major dynamics in making a shift away from behavioral concerns. The middle school should continue to evaluate the attendance policy and establish a procedure to inform parents about their thoughts on resolving their concerns and solicit parental involvement. This plan should meet the needs of the entire school community.

Qualitative Strand Findings

Research Question 8: In response to the perceptions on effectiveness, fairness, and training/satisfaction of teachers after training (but before institution) in restorative discipline practices the focus group responded students needed more instruction on how to respond to situations that produce behavioral issues with other students and ways to respond to those issues. The focus group deemed communication was the key component in resolving behavioral difficulties and providing leadership roles for students who felt they did not have a voice within the school environment. Teachers stated, "We are in constant communication with other teachers about students." They reported, "Restorative practice circles were necessary for the reduction of tardies and absenteeism."

The focus group considered communication with all stakeholders was needed for the enhancement of a restorative communication narrative. Restorative communication should be built on the principle of relationships that emphasize the primary role of creating and maintaining relationships for all communicators. Positive communication builds positive relationships (Bledsoe, 2018). They reported the use of restorative circles did provide students the voice of respect they felt they deserved. The teachers commented, "Students have a voice in making decisions." The personal benefit of providing communication that is respectful to each other can reaffirm our inherent dignity (Bledsoe, 2018). This type of communication is what the focus

group was hoping restorative practices would provide to their students and all people involved in their educational growth.

The focus group also felt that all stakeholders should take responsibility for the ongoing health of the school climate. If stakeholders miss the mark and do not provide healthy communication through restorative practices, they will impact others and all their relationships with them.

Research Question 9: In response to the perceptions on effectiveness, fairness, and training/satisfaction of teachers after a semester of instituting restorative discipline practices the focus group believed restorative practices gave cohesiveness not just to the students but the staff. The teachers stated, "Everyone is on the same page with circles including families." Emphasis was placed on providing a program where all stakeholders were together in their decision-making endeavors. The focus group expressed that they were holding their students to a higher standard towards their education and behavior while providing goals for future improvement in school climate. The group reported that the students felt their voices were being heard and the reinforcement of social skills including mutual respect was increasing. They stated, "Everyone is working together for the same goal." With an increase in mutual respect, the students should start to develop work and career-ready attitudes along with improved relationships between and among students, teachers, staff, and administrators.

Findings confirm that restorative practice programs and interventions vastly impact school climate. Research indicates that the feelings of remorse created through restorative practice programs also lead to changes in student behavior (Alvis, 2015). Morrison (2006) found that school-based restorative practice programs are effective in decreasing bullying and other

harmful behavior. This is consistent with the qualitative research conducted within the focus group who voiced that the wrong-doers were being held accountable for the effects of their actions on others while providing minimization of disruption and distraction among their peer groups. One teacher stated, "I noticed our students not being as distracted and more on task during the class and while doing their in-class assignments."

Research Question 10: In response to having changed the teachers' perceptions on effectiveness, fairness, and satisfaction regarding restorative discipline practices in this time frame, the focus group felt that the restorative practice program had provided more language communication, more organization within the classroom expectations, and more conversations conducive to learning. At the beginning of the program, the focus group stated they were nervous about implementing a new plan but were hopeful the program would have positive results and an improvement in the school climate. They were thinking optimistically about the program and "We have belief in the program and need to work together to gain overall positive results." Upon implementation of the restorative practice program, the focus group felt restorative practices enhanced all stakeholder's sense of belonging and develop confidence in their ability to reach a mutual understanding while developing creative solutions to issues within their classroom. The group felt that trust and honesty were also built within the school environment. The focus group felt "This program has created lasting relationships and built trust to be able to handle conflicts constructively." Lastly, the focus group reiterated "We are glad this program will help bring all stakeholders together."

Limitations of the Study

Although steps were taken to ensure the quality of research during this given time frame, barriers remain that need to be addressed as part of this study. First, the results indicated the

short time frame for this study might have been inadequate because research states a "roll-out plan needs to plan trainings, parent involvement and the how-to of at least a 1-3 year plan" (Brummer, 2016, p. 4).

Another limitation of this study was that some teachers within the focus group did not have a knowledge base about restorative practices or programs before the training. Lack of knowledge on the subject by the teachers within the focus group may have skewed their responses due to an insufficient understanding of the topic. Follow-up trainings were provided to the focus group by the administration using PLC's for progress after the implementation of the program.

The study did not take into account that teachers had to give up their planning time, which prevented the teacher from addressing other educational subjects. Student absences and tardies also produce issues with providing a scheduled time for circles to be presented by the focus group. Restorative practices require individual training for teachers and school personnel that can be costly and time-consuming for many schools (McCluskey, 2008). Also, restorative practices require a whole-school approach, which is often challenging because it requires schools to adopt and enforce clear and specific standards that meet the goals of the program (McCluskey, 2008).

Conclusions

There were two general conclusions drawn from this study. The first conclusion is that the quantitative data showed there was no significant difference in discipline referrals, out-of-school suspensions, absenteeism, and chronic absenteeism in fall 2018 and fall 2019. The inschool suspensions, in-school and out-of-school suspensions, and alternative school showed fall 2018 was significantly less than fall 2019. These results indicate that the restorative practice

program needs to in place longer than one semester to see the benefits over an extended time period. While improvement was not seen immediately, this is typical of a large change in school discipline and attendance. The effort to improve and maintain good discipline and attendance is a long term, ongoing commitment. More time is required to adequately measure whether a program such as this could provide data showing a significant change. Researchers suggest that a change in mindsets toward punishment may take one to three years and a change in school climate might take up to three to five years (Evans & Lester, 2013), (Karp & Breslin, 2001).

The second conclusion is that the qualitative data found many positive results from the focus group of educators in the implementation of this restorative practice program. The focus group found many examples of meaningful exchanges and a program, which used the circle processes by respectively repairing the students who were involved in the infractions discussed in the study. The circle process allows people to tell their stories and offer their perspectives (Pranis, 2005). Circles offered the students an alternative to contemporary meeting processes that are often relied upon as a win-lose positioning and argument (Roca, Inc., n.d.). The focus group of teachers reported that their students talked about using the circle process to work through their conflicts because it was intrinsically democratic and allowed all voices to be heard. The teachers indicated the students felt a sense of empowerment to be able to initiate and be a part of the process. The focus group believed all of the participants did take responsibility for maintaining the integrity of the group while building confidence in their ability to reach a mutual understanding and develop creative solutions to their problems. Student responses revealed to the focus group of teachers the importance of trust and fairness and how that, in turn, played an important role with the focus group towards their satisfaction with the restorative practice

program. Also, the teachers reported the students developed good listening and speaking skills while minimizing aggressive behaviors.

Although the data did not show an improvement in some areas, the focus group and administration found that through the enhancement of a restorative practice program and providing communication with all stakeholders, a healthy and successful gauge of the program's strength was a present factor. The focus group deemed the restorative practice program increased communication among students and teachers, developed cohesiveness with all stakeholders, and gave students a voice. They felt the climate of the school improved and many students developed leadership roles within the school. As the school moves forward with the program, providing communication between the school and community is a vital component. As changes are made with school board policies, data will need to be updated for an alternative school, in-school and out-of-school placements. The effectiveness of this restorative practice program will need more time and research to provide the necessary data needed to see the success of this program.

Recommendations

This study was designed to discover if a restorative practice program implemented in a low-income middle school would have an effect on discipline referrals, discipline, in-school suspensions, out-of-school suspensions, alternative school, absenteeism, chronic absenteeism and teacher's perspective before and after the implementation of a restorative practice program. The quantitative data collected reflected no significant difference notation on discipline, discipline referrals, out-of-school suspensions, absenteeism, and chronic absenteeism upon implementation of a restorative practice program. The fall 2018 semester was significantly less than fall 2019 in in-school and out-of-school suspensions, in-school suspensions and alternative school

placements upon implementation of a restorative practice program. The focus group of teachers felt that the restorative practice program along with the interventions put in place vastly impacted the overall school climate. A recommendation should include during the beginning of each school year a scheduled restorative practice instructional program should be designed and made available for all stakeholders. This program will provide empowerment throughout the implementation of a restorative practice program and engage all parties involved in the decision-making process. Based on the results of this study, further research with practical applications of a restorative practice program on attendance and discipline should be studied. Parental meetings with educators should be scheduled during the first two weeks of school to address attendance and discipline policies. After scheduled meetings with parents and educators, an on-going training program should be established with peer coaching for all staff to provide the necessary support and early intervention needed for a successful restorative practice program. The administration should provide opportunities for a school wide flexible schedule for circles and offer training for students and families to introduce a restorative practice approach.

The findings of this study will not be conclusive until more time and a widespread distribution of the findings can be administered and measured with validity. After that point, the findings can be used to help promote social and emotional growth with a positive school climate for all stakeholders.

Recommendations for Practice

Based on the findings of this study, the following recommendations are made for practice:

1. The school should implement a restorative practice professional development program for all staff members and make the initial investment in building

- communication between the school and community resources to establish a restorative practice program that lasts and helps ease the burdens of implementation.
- 2. The current employees of the school should be responsible for improving school climate by distributing tasks to advocate restorative practice programs with families at times when restorative action outside classroom activities is needed. The administration of the school should be responsible for creating professional learning opportunities, providing peer coaching, and ensuring instruction that provides the requirements for a restorative practice program to work within the school community.
- 3. A designated position for the administration of the restorative practice activities should be assigned and monitored by administrators to have assessment mechanisms for quantitative and qualitative review, and accountability of the procedures in place.
- 4. Administration and staff members should conduct readiness assessments to develop a model for change and a timeline for implementation for the entire school.
- 5. The school along with the district should support the restorative practice professional development viewpoint and help bridge the responsibilities of the staff to form a collaborative environment where all stakeholders know what is needed to support all students.
- 6. The school administration should allow more time before making a decision on the effectiveness of a restorative practice program.

Recommendations for Future Research

This review shows a need for more research on the use and implementation of restorative practices within this school. Based on the data and interviews with the focus group, the evidence supports the concept of relationship building along with all levels of the school community as a means to prevent and intervene in discipline referrals, discipline, absenteeism, and chronic absenteeism. In addition to decreasing discipline referrals and absenteeism and a positive student and adult relationship being fostered, it is hopeful that school suspensions would be fewer in number due to the process. In the future, a research program would need to be implemented and a hypothesis developed to see if the change in suspensions and alternative school placements are possible. Based on the results of this study, it is clear that further research with the practical application of restorative practice in all grade levels is necessary for a longer period of time such as three years to allow for the change process to be cultivated. There is a lack of studies on the long-term effects of restorative practices including recidivism, community change, and the psychological impact on the student (Gumz & Grant, 2009). There is a need for studies using multiple schools along with data measurement to examine the short and long term effects of restorative practice initiatives. The demographics of the school along with the students and how it relates to the implementation of a restorative practice program's success needs further research. This research could be conducted by using follow up interviews with all stakeholders and observing behavioral and absentee trends over a set period of time. Although there are few empirical studies in the area of restorative practice and its use in schools, studies that exist suggest that it improves the school environment and enhances learning opportunities which may lead to a decrease in behavioral and absentee problems (Chmelynski, 2005). As the teachers

reflected in their interviews, they felt better school experiences would happen when proactive methods such as restorative practices were used in conjunction with responsive measures.

The information from this study and additional research on this topic would provide administrators with the benefits of using restorative practices in their school districts. By learning about the teachers' perceptions will help school districts to recognize and work through any problems that would prohibit the successful implementation of a restorative practice program. More documentation and data on these topics for a longer period should be made available to see the full benefits of advocating for changes in the handling of school discipline referrals, discipline, absenteeism, and teacher perceptions on the matter. This information could greatly impact the future of the students and all stakeholders.

REFERENCES

- Alvis, M. (2015). Teachers' Perceptions About Using Restorative Practice Based Programs in Schools.
- American Psychological Association Zero Tolerance Task Force, (2008). Are Zero Tolerance Policies Effective in the Schools? An Evidentiary Review and Recommendations:

 *American Psychologist, 63(9), 852-862.
- Anfara, V. A., Jr., Andrews, P.G., Hough, D. L., Mertens, S. B., Mizelle, N. B., & White, G. P. (2003). *Research and resources in support of This We Believe*. Westerville, OH: National Middle School Association.
- Ashley, J., & Burke, K. (2009). *Implementing restorative justice: A guide for schools*. Chicago, IL: Illinois Criminal Justice Information Authority.
- Arnold, J. (2014). What is Discipline? Retrieved from https://www.plough.com/en/topics/community/education/what-is-discipline
- Augustine, C., Engberg, J., Grimm, G., Lee, E., Wang, E., Christianson, K., & Joseph, A. (2018). An Evaluation of the Impact of Restorative Practices in a Mid-Sized Urban School District, Can Restorative Practices Improve School Climate and Curb Suspensions?
- Balfanz, R., & Byrnes, V. (2012). The Importance of Being There: A Report on Absenteeism in the Nation's Public Schools, *Johns Hopkins University School of Education*, 1-46.
- Bazemore, G. (2000). Restorative juvenile justice: Repairing the harm of youth crime.

 Restorative Justice and Relational Rehabilitation. Criminal Justice Press. Monsey, N.Y:

 Criminal Justice Press.

- Bell, A. J., Rosén, L. A., & Dynlacht, D. (1994). Truancy intervention. *Journal of Research & Development in Education*, 27(3), 203–211.
- Bitel, M. (2005). National Evaluation of the Restorative Justice in Schools Programme. London, UK: Youth Justice Board for England and Wales.
- Bledsoe, W. (2018). The 7 Principles of Restorative Communication. Retrieved from https://restorativeway.consulting/2018/10/the-7-principles-of-restorative-communication/
- Boucaud, A. (2019). Dissertation Spotlight: Examining the Relationship between Restorative Practices and School Climate. Retrieved from www.education.cu-portland.edu/dissertations-spotlihgt
- Boyes-Watson, C. (2005). Seeds of change: Using peacemaking circles, to build a village for every child. *Child Welfare*, 191-208.
- Braithwaite, J. (2007). *Crime, shame and reintegration*. Cambridge: Cambridge University Press.
- Brummer, J. (2016). 5 Reasons Implementation of Restorative Practices Fails in Schools.

 Retrieved from http://www.joebrummer.com/2016/02/28/5-reasons-implementation-of-restorative-practices-fails-in-schools/
- Catapano, J. (2019). Restorative Justice in School Discipline. Retrieved from www.teachhub.com/restorative-justice-school
- Chmelynski, Carol. (2005). Restorative Justice for Discipline with Respect. *Education Digest:*Essential Readings Condensed for Quick Review, 71(1), 17-20.
- Choi, J. J., Green, D. L., & Gilbert, M. J. (2011). Putting a Human Face on Crimes: A

 Qualitative Study on Restorative Justice Processes for Youths. *Child and Adolescent*

- Social Work Journal, 28(5), 335–355.
- Cobb-Clark, D. A., Kassenboehmer, S. C., Le, T., Mcvicar, D., & Zhang, R. (2015). Is there an educational penalty for being suspended from school? *Education Economics*, 23(4), 376–395.
- Cole, Nicki Lisa, Ph.D. (2020). Understanding the School-to-Prison Pipeline.

 Retrieved from https://www.thoughtco.com/school-to-prison-pipeline-4136170
- Corville-Smith, J., Ryan, B., Adams, G., & Tom, D. (1998). Distinguishing Absentee Students from Regular Attenders: The Combined Influence of Personal. Family, and School Factor. *Journal of Youth Adolescence*, 27(5), 629–640.
- Creswell, J.W. (2009). Research design: Qualitative, quantitative, and mixed methods approaches. Thousand Oaks, CA: Sage.
- Davis, M. (2015). Restorative Justice: Resources for Schools. Retrieved from https://www.edutopia.org/blog/restorative-justice-resources-matt-davis
- Dalporto, Deva. (2013). Restorative Justice: A Different Approach to Discipline, We are

 Teachers. Retrieved from http://www.weareteachers.com/restorative-justice-a-different-approach-to-discipline/
- Daly, K. & Proietti-Scifani, G. (2017). Gendered Violence and Restorative Justice: The Views of New Zealand Opinion Leaders. Retrieved from www.

 griffith.edu.au/_data/assets/pdffile/0024
- Drewery, W. (2004). Conferencing in schools: punishment, restorative justice, and the productive importance of the process of conversation. *Journal of Community & Applied Social Psychology*, 14(5), 332–344.
- Durham District School Board Safe Schools. (2012). Restorative Practice in the Durham

- District School Board. Retrieved from http://docplayer.net/47186061-Restorative-practice-in-the-durham-district-school-board-safe-schools-presentation-ddsb-fall-2012.html
- Evans, K., & Vaandering, D. (2016). The Little Book of Restorative Justice in Education: Fostering Responsibility, Healing, and Hope in Schools. Good Books.
- Evans, K. & Lester, J. (2013). Restorative justice in education: What we know so far. *Middle School Journal*, 44(5), 5-63.
- Fabelo, T., Thompson, M., Plotkin, M., Carmichael, D., Marchbanks, M., III, & Booth, E.
 (2011). Breaking Schools' Rules: A Statewide Study of How School Discipline Relates to Students' Success and Juvenile Justice Involvement, New York: Council of State Governments.
- Fronius, T., Persson, H., Guckenburg, S., Hurley, N., & Petrosino, A. (2016). *Restorative Justice in U.S. Schools: A Research Review*, San Francisco, Calif.: WestEd Justice & Prevention Research Center.
- Gill, P., Stewart, K., Treasure, E., & Chadwick, B. (2008). Methods of data collection in qualitative research: interviews and focus groups. Retrieved from https://www.nature.com/articles/bdj.2008.192#Sec6
- Gonzalez, Thalia (2012). Keeping Kinds in Schools: Restorative Justice, Punitive Discipline, and the School to Prison Pipeline, *Journal of Law and Education*, 41(2), 281-335.
- Gregory, A., Clawson, K., Davis, A., & Gerewitz, J. (2016). Restorative Practices to Transform

 Teacher-Student Relationships and Achieve Equity in School Discipline, *Journal of Educational and Psychological Consultation*, 26 (4), 325-353.

- Grossi, P., & Santos, A. (2012). Bullying in Brazilian schools and restorative practices.

 Canadian Journal of Education, 35(1), 12-136.
- Gumz, E.J., & Grant, C. L. (2009). Restorative Justice: A Systematic Review of the Social Work Literature. *Families in Society*, 90(1), 119-126.
- Haft, W. (2000). More than zero: The Cost of Zero Tolerance and the case for Restorative Justice in Schools. *Denver University Law Review*, 77(4), 795-812.
- Haiman, P. (1998). Time out to correct misbehavior may aggravate it instead, *The Brown University Child and Adolescent Behavior Letter*, 14(10), 1.
- Haley, N.A. (2000). In-school literacy extension: Beyond in-school suspension. *Journal of Adolescent & Adult Literacy*, 43(7), 654-661.
- Hargreaves, A. E. (1997). *Rethinking Educational Change with Heart and Mind. 1997 Ascd Yearbook*. Association for Supervision and Curriculum Development.
- Hopkins, Belinda. (2003). *Transforming Conflict Centre for Restorative Justice in Education*.

 Mortimer Hill.
- Hopkins, B. (2004). *Just schools a whole school approach to restorative justice*. London: J. Kingsley Publishers.
- Howard, Paul. (2009). Restorative Practice in Schools. Retrieved from www.cfbt.com
- Hughes, A. S. (n.d.). Mixed Methods Research. Retrieved from https://www.psychologicalscience.org/observer/mixed-methods-research
- Jain, S., Henrissa B., Brown, M., & Preety. K. (2014). Restorative Justice in Oakland Schools:
 Implementation and Impacts, An Effective Strategy to Reduce Racially Disproportionate
 Discipline, suspensions, and Improve Academic Outcomes. Retrieved from

- https://www.ousd.org/cms/lib/CA01001176/Centricity/Domain/134/OUSD-RJ%20Report%20revised%20Final.pdf
- Karp, D. & Breslin, B. (2001), Restorative justice in school communities. *Youth and Society*, 33(2), 249-272.
- Kaveney, K. & Drewery, W. (2011). Classroom meetings as a restorative practice: A study of teacher's responses to an extended professional development innovation. *International Journal on School Disaffection*, 8(1), -5-12.
- Kidde, J., & Alfred, R. (2011). Restorative Justice Center for Healthy Schools. Retrieved from http://www.achealthyschools.org/schoolhealthworks/assets/118-restorative-justice.pdf
- Kim, W. C., & Mauborgne, R. (2003). Fair process: managing in the knowledge economy,

 Ma.: Harvard Business School Pub. Co.
- Knoff, H. (2015). Retrieved from http://communities.nasponline.org/blogs/howard-knoff/2015/03/15/restorative-practices-and-reducing-suspensions-the-numbers-just-dont-add-up
- Lane, D. (2005). The building of relationships. *International Journal of Learning*, 12(5), 41-50.
- Leech N, Onwuegbuzie A, (2008) A typology of mixed methods research designs, *Quality and Quantity*, 43(2), 265-275.
- Lewington, J. (2016). The Power of Restorative Practice, Education Canada, 16-19.
- Macready, T. (2009). Learning social responsibility in schools: A restorative practice, *Educational Psychology in Practice*, 25(3), 211-220.

- McCluskey, G., Lloyd, G., Kane, J., Riddell, S., Stead, J., and Weedon, E. (2008). Can

 Restorative Practices in Schools Make a Difference? *Educational Review*, 60(4), 405-417.
- McCold, P. (2003). A survey of assessment research on mediation and conferencing.

 Repositioning restorative justice. Devon, UK: Willan.
- McMorris, B.J., Beckman, K.J., Shea, G., and Eggert, R.C., *A pilot program evaluation of the family and Restorative Conference Program*. Retrieved from www.legalrightscenter.org/uploads/2/5/7/3/25735760/lrc-umn-report-final.pdf
- Mendez, L.M, Knoff, H.M., & Ferron, J.M. (2002). School demographic variables and out of school suspension rates: A quantitative and qualitative analysis of a large, ethnically diverse school district. *Psychology in the Schools*, *39*(3), 259-277.
- Meyer, L.H., and Evans, I.M. (2012). *The School Leader's Guide to Restorative School Discipline*, Thousand Oaks, CA.
- Mirsky, L. (2003). SaferSanerSchools™: Transforming School Culture with Restorative

 Practices. Retrieved from https://iirp.edu/news/safersanerschools-transforming-schoolculture-with-restorative-practices
- Morrison, B. (2006). School bullying and restorative justice: Toward a theoretical understanding of the role of respect, pride, and shame. *Journal of Social Issues*, 62(2), 371-392.
- Morrison, B., & Vandering, D. (2012). Restorative justice: Pedagogy, praxis, and discipline. *Journal of School Violence*, 11(2), 138-155.
- National Center for Education Statistics. (2009). *The Condition of Education 2016, NCES 2016-144,* Washington D.C.: U.S. Department of Education.

- National Council on Teacher Quality. (2014). Roll Call: The Importance of Teacher

 Attendance, Executive Summary.
- Norris, Alexis. (2009). Gender and Race Effects of a Restorative Justice Intervention on School Success, paper presented at the annual meeting of the American Society of Criminology, Philadelphia, Pa.
- Okonofua, Jason A., David Paunesku, and Gregory M. Walton. (2016). Brief Intervention to Encourage Empathic Discipline Cuts Suspension Rates in Half Among adolescents, Proceedings of the National Academy of Sciences of the United States of America (PNAS), 113(10), 5221-5226.
- Ontario Ministry of Education. (2006). Safe Schools Policy and Practice: An Agenda for Action, Toronto, Ontario, Canada.
- Porter, A. (2007). Restorative Practices in Schools: Research reveals Power of Restorative Approach, Part 2. Retrieved from http://www.lirp.educ/eforum-archive/4366-restorative-practices-in-schools: research-reveals-power-of-restorative-approach-part-ii
- Pranis, K. (2005). *The little book of circle processes: a new/old approach to peacemaking*. Intercourse, PA: Good Books.
- Perez, J. (2015). Teachers complain about revised CPS discipline policy: Retrieved from https://www.chicagotribune.com/news/ct-cps-discipline-concerns-met-20150225-story.html
- Raebeck, B. (1993). Beyond the dunce cap. The Executive Educator, 15(4), 26-28.
- Restorative Practices Working Group. (2004). Restorative Practices: Fostering Healthy

 Relationships and promoting Positive Discipline in Schools: A Guide for Educator.

 Retrieved from www.otlcampaign.org/restorative-practices

- Riestenberg, N. (2002). Restorative measures in schools: Evaluation results. Minneapolis, MN.
- Riestenberg, N. (2003). Restorative Schools Grants Final Report, January 2002-June 2003:

 A Summary of the Grantees' Evaluation, Roseville, Minn.: Minnesota Department of Education.
- Riley, B. (2018). A Better Approach to School Discipline. *Principal.* 97(3)
- Roca, Inc. (n.d.) Peacemaking circles: A process for solving problems and building community.

 Retrieved from

 http://ocs.archchicago.org?LinkClick.aspx?fileticket_YyvmyeF8KDg%tabid=996

 &language=en-US
- Rumberger, Russell W., and Daniel J. Losen. (2017). The Hidden Costs of California's Harsh

 School Discipline: And the Localized Economic Benefits from Suspending Fewer High

 SchoolStudents, Los Angeles, Calif.: Center for Civil Rights Remedies at the UCLA

 Civil Rights Project.
- Safe Schools Action Team. (2008). Shaping a culture of respect in our schools: Promoting safe and healthy relationships. Toronto, Ontario, Canada: Queen's Printer for Ontario.
- Simson, D. (2012). Restorative Justice and Its Effects on (Radically Disparate) Punitive School Discipline, Stanford, Calif.: Stanford Law School.
- Shah, N. (2012). Restorative Practices': Discipline but Different, *Education Week, 32*(8), 14-15.
- Skiba, R.J., Peterson, R.L., & Williams, T. (1997). Office referrals and suspension:Disciplinary intervention in middle schools. Education and Treatment of Children,20(3), 295-315.

- Skiba, Russell J. (2014). "The Failure of Zero Tolerance," *Reclaiming Children and Youth*, 22 (4), 27-33.
- Skiba, R. J., Trachok, M., Choong-Geun Chung, Baker, T., Sheya, A., & Hughes, R. (2014).

 Where Should We Intervene? Contributions of Behavior, Student, and School

 Characteristics to Suspension and Expulsion, *Closing the School Discipline Gap:*Research for Policymakers, New York: Teachers College Press.
- Skiba, R. (2006). Zero tolerance, suspensions, and expulsion: Questions of equity and effectiveness, *Handbook of classroom management: Research, practice, and Contemporary Issues*, 1063-1092.
- Solter, A. (1998). Why Do Children "Misbehave? Aware Parenting Institute: Retrieved from http://www.awareparenting.com/misbehav.htm
- Strickland, V. P. (1998). *Attendance and Grade Point Average A Study*. Place of publication not identified: Distributed by ERIC Clearinghouse.
- Summer, M., Silverman C., & Frampton, M. (2010). School-Based Restorative Justice as an alternative to Zero-Tolerance Polices: Lessons from West Oakland, Berkeley:

 University of California
- Tennessee Department of Education. (2019). Retrieved from https://www.tn.gov/education/student-support/chronic-absenteeism.html
- Tomczuk, K. (2000(. Prevention, not punishment. American School Board, 187(5), 60-61
- Turpin, T. & Hardin, D.T. (1997). A case study of an in-school suspension program in a rural high school setting. *Research in the Schools*, 4(2), 57-63.
- U.S. Department of Education. (2016). The 2013-14 Civil Rights Data Collection (CRDC):

- First Look, Washington, D.C.
- U.S. Department of Education (n.d.). Retrieved from http://www2.ed.gov/programs/titleiparta/index.html
- Varnham, S. (2005). Seeing things differently: Restorative justice and school discipline. *Education & The Law*, 17(3), 87-104.
- Vincent A. Jr., Evans, K., & Lester, J. (2013). Restorative Justice in Education: What We Know So Far, *Middle School Journal*, 44(5), 57-63.
- Wachtel, T. (2016). *Defining Restorative*. Retrieved from https://www.iirp.edu/pdf/Defining-Restorative.pdf
- Walgrave (nd.), Repositioning Restorative Justice. Devon, UK: Willan.
- Winslade, J. & Williams, M. (2012). Safe and Peaceful Schools: Addressing conflict and eliminating violence. Thousand Oaks, CA: Corwin.
- Zaslaw, J. (2009). A restorative approach to resolving conflict. *Principal*, 48-49.
- Zehr, H. (2002). The little book of restorative justice. Intercourse, PA: Good Books.
- Zehr, H., Stutzman A., MacRae, A., & Pranis, K. (2015). The Big Book of Restorative Justice:

 Four classic justice & peacebuilding books in one volume. New York, NY: Good

 Books.
- Zheng, S., & De Jesus, S. (2018). Effects of restorative practices on suspended pupils' school engagement, academic achievement, and post-secondary destinations. Toronto, Ontario, Canada: Toronto District School Board

APPENDICES

APENDIX A



Date: July 19, 2019

From: The Institutional Review Board (IRB) at Milligan College

Re: A Restorative Practice Program's Effects on Discipline Referrals, Absences, and Teacher Perspectives

Submission type: Revised Submission

Dear Nancy,

On behalf of the Milligan College Institutional Review Board (IRB), we are writing to inform you that your study *A Restorative Practice Program's Effects on Discipline Referrals, Absences, and Teacher Perspectives* has been approved as expedited. This approval also indicates that you have fulfilled the IRB requirements for Milligan College.

All research must be conducted in accordance with this approved submission, meaning that you will follow the research plan you have outlined here, use approved materials, and follow college policies.

Take special note of the following important aspects of your approval:

- Any changes made to your study require approval from the IRB Committee before they can be implemented as part of your study. Contact the IRB Committee at IRB@milligan.edu with your questions and/or proposed modifications.
- If there are any unanticipated problems or complaints from participants during your data collection, you must notify the Milligan College IRB Office within 24 hours of the data collection problem or complaint.

The Milligan College IRB Committee is pleased to congratulate you on the approval of your research proposal. Best wishes as you conduct your research! If you have

faculty advisor if appropriate on the communication.
Regards,
IRB Committee

Milligan College, Tennessee 37682 • (423) 461-8700 • www.milligan.edu Changing lives, shaping culture – a commitment to Christian leadership

APENIX B

Principal Permission to Conduct Study

To:
From: Nancy Harris Parker
Date:
Subject: Principal Permission to Conduct Study
I am currently an Ed.D. candidate attending Milligan College. The research for my doctoral dissertation involves the use of restorative practices in a Title I middle school in East Tennessee. The proposed dissertation will include the effects of restorative practices on student behavior, grade enhancement, and teacher satisfaction.
The purpose of this study will be to use discipline referrals, grade point averages, and teacher perspectives before and after the use of restorative practices. Teacher surveys will be collected at the beginning and end of the fall semester. All information will be kept confidential and teacher surveys will be anonymous.
This study will require teacher participation and provide feedback to educators about the changes in curred utilizing the restorative practices program. It will correlate and chart changes in discipline patterns and notate the relationship process of the before and after pertaining to grades through the use of restorative practices. The study will require two meetings to conduct surveys of the eighth grade faculty at the beginning of the school year, August 2019, and again at the beginning of the second semester of the school year, January, 2020. The meetings will be held in the high school/middle school library for approximately 15 -20 minutes during scheduled faculty meetings. Information regarding discipline referrals and grades will be gathered by school administrative staff and all information will be kept confidential.
This project will be conducted under the supervision of, Principal of Middle School. Dr. Mark Dula will be overseeing my dissertation work at Milligan College.
The possible benefits of this study for your school are developing a restorative practice program that has already been establish suggests positive outcomes in student discipline including zero tolerance and alternative school, enhancement of grades, and teacher perspective regarding school climate. I will be collecting data to evaluate how a new program will benefit the school.
I respectfully request your permission to conduct this research study at Middle School as part of my doctoral dissertation.

Upon completion of the study, I will provide the copy of the research report upon request. Thank you for your time and consideration in this matter. form to:	
Nancy Harris Parker	
Your signature indicates that you have read and understood you willing agree for me to participate in the restorative proceduct this study, and that you have received a copy of the	actice program, to
Yours sincerely,	
Nancy Harris Parker	
I hereby consent to my school district's participation in the	e research described above.
School District	
Principal Signature	
Principal Print	
Date	

APENDIX C

Superintendent Permission to Conduct Study

To:
From: Nancy Harris Parker
Date:
Subject: Superintendent Permission to Conduct Study
I am currently an Ed.D. candidate attending Milligan College. The research for my doctoral dissertation involves the use of restorative practices in a Title I middle school in East Tennessee. The proposed dissertation will include the effects of restorative practices on student behavior, grade enhancement, and teacher satisfaction.
The purpose of this study will be to use discipline referrals, grade point averages, and teacher perspectives before and after the use of restorative practices. Teacher surveys will be collected at the beginning and end of the fall semester. All information will be kept confidential and teacher surveys will be anonymous.
This study will require teacher participation and provide feedback to educators about the changes incurred utilizing the restorative practices program. It will correlate and chart changes in discipline patterns and notate the relationship process of the before and after pertaining to grades through the use of restorative practices. The study will require two meetings to conduct surveys of the eighth grade faculty at the beginning of the school year, August 2019, and again at the beginning of the second semester of the school year, January, 2020. The meetings will be held in the high school/middle school library for approximately 15 -20 minutes during scheduled faculty meetings. Information regarding discipline referrals and grades will be gathered by school administrative staff and all information will be kept confidential.
This project will be conducted under the supervision Principal of Dr. Mark Dula will be overseeing my dissertation work at Milligan College.
The possible benefits of this study for your school are developing a restorative practice program that has already been establish suggests positive outcomes in student discipline including zero tolerance and alternative school, enhancement of grades, and teacher perspective regarding school climate. I will be collecting data to evaluate how a new program will benefit the school.
I respectfully request your permission to conduct this research study at Middle School as part of my doctoral dissertation.

Upon completion of the study, I will provide the Department of Education with a copy of the research report upon request.
Thank you for your time and consideration in this matter. Please sign and return one copy of this form to:
Nancy Harris Parker
Your signature indicates that you have read and understood the information provided above, that you willing agree for me to participate in the restorative practice program at, to conduct this study, and that you have received a copy of this form.
Yours sincerely,
Nancy Harris Parker
I hereby consent to my school district's participation in the research described above.
School District
Superintendent Signature
Superintendent Print
Date

APPEDIX D

CONSENT FORM

You are invited to take part in a research study of "A Restorative Practice Program's Effects on Discipline Referrals, Absences, and Teacher Perspectives". The researcher is inviting a focus group of teachers to be in the study. This form is part of a process called "informed consent" to allow you to understand this study before deciding whether to take part. This study is being conducted by a researcher named Nancy Harris Parker who is a doctoral student at Milligan College.

Background Information:

The purpose of this study is to determine if there is a significant difference in the number of student discipline referrals, out-of-school suspensions, in-school suspensions, alternative school placements, chronically absent students and absences in general following the institution of restorative discipline practices. The study will also determine the perceptions of teachers on the effectiveness, fairness, training and satisfaction after training and implementing restorative practice within the school framework.

Procedures:

If you agree to be in this study, you will be asked to:

• Interview and discuss answers to prearranged questions

Voluntary Nature of the Study:

This study is voluntary. Everyone will respect your decision of whether or not you choose to be in the study. No one at _______, Department of Education, or Milligan College will treat you differently if you decide not to be in the study. If you decide to join the study now, you can still change your mind later. You may stop at any time.

Risks and Benefits of Being in the Study:

Being in this study will not pose risk to your safety or wellbeing.

This study into restorative practices in a Title I educational institution can give teachers more discretion to implement restorative punishment for nonviolent behavior and develop better relationships with their teachers. It could be that better student-to-staff relationships will lead to improved student behaviors and attendance.

Payment:

There will be no remuneration for this study.

n	•			
ν	rı	va	Λī	7•
_		v a	v	•

Any information you provide will be kept confidential. The researcher will not use any personal information for any purposes outside of this research project. The researcher will not include your name or anything else that could identify you in the study. Data will be kept secure by Nancy Harris Parker. Data will be kept for a period of at least 5 years, as required by Milligan College.

College.	period of at least 5 years, as required by Milligan
Contacts and Questions:	
	if you have questions later. You may contact the you want to talk privately about your rights as a lege Institution Review Board at
	I understand the study well enough to make a below, "I consent". I understand that I am agreeing
Printed Name of Participant	
Date of consent	
Participant's Signature	
Researchers' Signature	

APPENDIX E

Interview Questions for Focus Group of Educators August 2019

- 1. Thinking back over this past school year, tell me some ways the school environment was conducive to learning.
- 2. Thinking back over this past school year, give me examples of the main issues regarding discipline and attendance in this school climate.
- 3. Describe how you feel a restorative practice program will have an effect on the school climate and teacher satisfaction.
- 4. Describe how you feel a restorative practice program will have an effect on discipline and attendance in your school.
- 5. Tell me some ways the majority of the staff believe restorative practice can improve student behavior.
- 6. Describe how the majority of the staff believe restorative practice can improve student attendance.
- 7. Tell me some ways the staff is familiar with the purpose and methods of restorative practices.
- 8. Describe how the staff feels about implementing restorative practices with the majority of the students of _____ Middle.
- 9. Describe how you feel the school climate will improve with the result of restorative practices.
- 10. Give me some examples how the staff believes restorative practices will improve the overall school culture.
- 11. In your experience what leads you to believe that students will handle conflict with other students as a result of restorative practices.
- 12. In your experience what leads you to believe that the staff will handle conflict with students as a result of restorative practices.

APPENDIX F

Interview Questions for Focus Group of Educators

December 2019

- 1. Thinking back over the fall semester, how has the school environment been more conducive to learning after implementing restorative practices.
- 2. Thinking back over the fall semester, name issues regarding discipline and attendance which have improved after implementing restorative practices.
- 3. Tell me some ways the school climate and teacher perceptions have improved upon implementation of restorative practices.
- 4. Give me an example of how restorative practices have improved discipline and attendance in your school.
- 5. What aspects of your experience with restorative practices leads you to believe the majority of the staff now believes restorative practices has improved overall student behavior.
- 6. What about your experience leads you to believe the majority of the staff now believes restorative practices has improved overall student attendance.
- 7. Tell me some ways the staff is more comfortable with the purpose and methods of restorative practices.
- 8. Give me some ways the staff is more confident implementing restorative practices now with the majority of the students of _____ Middle.
- 9. Tell me some ways the school climate has improved with the result of restorative practices.
- 10. Give me some examples of how the staff believes restorative practices has improved the overall school culture.
- 11. Describe how you feel the students can handle conflict with other students as a result of restorative practices.
- 12. Describe how the staff feels they can handle conflict with students as a result of restorative practices.